Funded by the European Union within the 7th Framework Programme, Grant Agreement 603378. Duration: February 1st, 2014 – January 31th, 2018





Deliverable 5-C: Report on legacy and tipping points in large rivers

Lead beneficiary: University of Natural Resources and Life Sciences Vienna (BOKU)

Authors: Wolfram Graf (BOKU), Florian Pletterbauer (BOKU), Gertrud Haidvogl (BOKU), Patrick Leitner (BOKU), Gerben van Geest (Deltares), Tom Buijse (Deltares), Sebastian Birk (UDE)

Due date of deliverable: **Month 24** Actual submission date: **Month 25**

Dissemination Level				
PU	Public	Х		
PP	Restricted to other programme participants (including the Commission Services)			
RE	Restricted to a group specified by the consortium (including the Commission Services)			
CO	Confidential, only for members of the consortium (including the Commission Services)			



Content

Content	2
Non-technical summary	4
Introduction	5
Objective	5
Background	5
Large river functioning	7
The history of drivers and pressures in European large rivers	9
The case study rivers	9
Qualitative historical timeline of occurrence of drivers and pressures	10
Differences between rivers	10
Flood protection and agriculture	11
Navigation	11
Construction of dams	12
Pressure effects on hydromorphological processes and ecology	12
Effects of deforestation	12
Effects of embankments	12
Effects of pressures related to navigation	13
Construction of dams	14
Conclusions on the stressor history in the case studies	15
Selected quantitative stressor timelines in large European rivers	16
Hydromorphological changes	16
Channelization	17
Damming	
Pollution	20
Navigation & Neozoa	21
Navigation	21
Neozoa	24
Large river fauna	
Changes in the Plecoptera fauna of large European rivers	
Definition of species of interest	



Sources for species records of Plecoptera	30
Spatial and temporal units of analyses	30
Results & Discussion	31
Fish assemblages of the Danube	36
Materials	36
Results & Discussion	37
Based on the different documents and sources finally used several trends of fish asse	emblage
development can be described for the Danube:	37
Synthesis	42
References	48
Appendix	56



Non-technical summary

Large rivers have always been in the focus of human attention. Ancient and modern civilisations have arisen, prospered and dwindled on their banks, leaving us with the myths and legends that their waters provoked. Large rivers are of major economic relevance as providers of substantial services like, most notably drinking water, food, energy and transport. Large rivers have been altered since centuries and have undergone dramatic human-induced changes. Thus, large rivers are among the most stressed ecosystems worldwide.

In large rivers the stability as well as the disturbance amplitude trigger integral ecosystem functions and determine where and when which habitats are available within a temporal context. Thus, the basis for a sufficient understanding on the effects of stressors and their role in faunal changes strongly depends on the understanding of the ecosystem processes and their interlinkages which determine ecosystem functioning. This deliverable addresses the historical development of stressors in large European rivers and their legacy on the faunal elements, which have been typical in those rivers. Nowadays rare potamontypic Plecoptera, aquatic insects, have been chosen to act as umbrella-species reflecting ecosystem health.

The analyses showed a common development of the major stressors damming, navigation and neozoa in large rivers during the second half of the 20th century, which can be observed for other economical and societal factors too. The major regulation with corresponding channelization effects have been already finished at the beginning of the 20th century. However, those channelization effects have been emphasised by measures related to damming and to the improvement of large river navigability.

We found a considerable shrink in the distribution area of selected indicator species, namely of the aquatic insect order Plecoptera, the stoneflies. The analyses identified few refugia in Central France (Loire, Allier), Austria/Hungary (Raba, Lafnitz) and Hungary/Romania (Tisza) where a combination of several species still can be found after 1990. These systems show some communalities like natural discharge and sediment regimes. Even though neozoans have invaded them, unbroken dynamic processes seem to lessen their negative effect on indigenous faunas. Losses are identified in many river systems, especially in Scandinavia and Spain. Even though the analyses are related to uncertainties due to data gaps, the general trend seems plausible.

Especially invasive species in consequence of inland navigation tremendously changed the faunal composition of aquatic insect assemblages. Biological reference communities are lost for most large rivers since long times and cannot be described empirically. To establish ecological integer ecosystems as demanded by the Water Framework Directive, the few river systems, which still sustain typical large river species, must serve as reference to restore hydrologically and sedimentologically dynamic habitats.



Introduction

Large rivers renege borders – neither political nor eco- or bioregional borders. Hence, they constitute transnational and –regional ecosystems. However, large rivers are among the most stressed ecosystems worldwide. Relevant stressors comprise channelization, damming, eutrophication, alteration of sediment transport, and neozoa, which are also found in smaller riverine ecosystems, and furthermore drainage and decoupling of floodplains as well as navigation, which are special to large river systems. As part of Work Package 5, task 5.2 deals with large rivers and their stressors on the European scale.

Objective

The task consists of two parts: (i) Legacy and tipping points, and (ii) Development of an assessment system for large European rivers. This deliverable is dedicated to the first part of the task which addresses the historical development of stressors on large European rivers and their legacy. Data from a wide range of European rivers will be summarised with emphasis on the Danube, the largest river in Central Europe, to give a qualitative as well as quantitative evaluation on the stressors and theirs effects on biological quality elements over time. The analyses will be related to different stressors and pressures like overfishing, pollution, channelization, dam construction, navigation, and invasive species. Finally, the stressors will be related to documented changes in the aquatic communities and ecosystem services which provides additional understanding of the legacy of large European rivers.

The findings will then feed into the second part of the task, which aims at the development of an assessment system for large European rivers as the currently existing assessment schemes have been mainly developed for small and medium-sized wadeable rivers, and lack full applicability for large rivers. The knowledge gained from both parts will help to assess the potential of recovery and provide guidance for prioritising restoration measures for large European rivers in the future.

Background

Independent of sectors, the development of socio-economy shows a uniform pattern on the global scale over time. Even tough, the global human uses last back for several hundreds of years, the intensity of human uses with according consequences for ecosystems tremendously speeded up since the 1950s (Steffen et al., 2015). The trend and acceleration of the anthropogenic developments like population growth, use of fossil fuels, global commerce, and industrial chemical processes combined and amplified with highly important implications for riverine ecosystems as well as large rivers. This relevance for riverine ecosystems is directly mirrorred by the increasing number of dams and increase of water use (Fig. 1). Accordingly, Fig. 1 also shows that the level of use steadily increased slowly till ~1950, followed by an exponential increase since then.



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 1: Socio-economic trends for twelve indicators from 1750 to 2010 from Steffen et al. (2015) including the river-relevant indicators 'Large Dams' and 'Water Use'

Large rivers have always been in the focus of human attention. Ancient and modern civilisations have arisen, prospered and dwindled on their banks, leaving us with the myths and legends that their waters provoked. Large rivers are of major economic relevance as providers of substantial services like, most notably drinking water, food, energy and transport (Demek et al., 2008; Kohl, 2010), well represented by a large number of European capitals situated next to large rivers.

Beside those benefits for human civilisation, large rivers are inherently linked to risks and uncertainties, especially due to flood risk. Therefore, mankind sought to control large rivers to gain better flood protection and to secure the provision of beneficial services. The waves of alterations on large rivers were strongly related to the technical abilities, like the invention of steam-engines, and the interest of socio-economy, like provision of goods to capitals.

Nearly all large rivers in Europe have undergone dramatic changes since decades, or even centuries. However, the documentation of these changes is still scattered and often related to sectoral perspectives and interests. In this report, the occurrence and development of various stressors through time should be described in a broader context. Starting with a synthesised



description of the history of drivers and pressures and followed by some detailed but representative examples of stressor developments in the 20th century. This more comprehensive view on the stressor development will be then linked up to tipping points of biological communities, i.e. on distribution patterns of large river Plecoptera species in Europe over time supplemented by the information on fish stocks in the Danube River. These analyses will be based on existing sources (i.e. reviewed literature). However, historical data often comprises inconsistencies and gaps. Thus, the identification of changes is mostly related to qualitative differences; where possible, quantifications were taken into account.

Large river functioning

Rivers can be classified by length, by discharge or by catchment area into size-classes (e.g. Tockner et al., 2009). The intercalibration exercise of the European Commission recognises rivers as large by a catchment size threshold exceeding 10,000 km². On a global scale discharge volume or river length are mostly used to build river size rankings and to identify the largest ones.

However, does such a classification or ranking reflect the ecosystem 'large river' properly? And in turn: Which processes and functions determine the characteristics of a large river ecosystem? And accordingly, are faunal elements, which are characteristic of large rivers, restricted in their occurrence to river systems with a defined catchment size? These questions are not directly related to the historical development of stressors but they have to be considered in the development of a functioning assessment system. Faunal elements as well as processes and functions characterising large rivers, are definitely not linked to size-indicators thresholds like length or catchment area as mentioned above.

Large rivers feature a multitude of habitats, which are linked to different processes, and thus, a diverse fauna under pristine conditions. Accordingly, some of those habitats and processes can be found in medium-sized rivers too. The knowledge on the temporal change of stressors and faunal elements should inform about the relevant ecosystem processes being impacted. In turn, it is possible to look for reference ecosystems where both the faunal elements and thus the intact ecosystem processes still occur. In summary, this implies the necessity to consider smaller river systems too.

Different ecological theories and concepts tried to synthesise the knowledge on processes and functions of riverine ecosystems and help to highlight characteristics, which can be expected in large rivers. The river continuum concept (Vannote et al., 1980) postulated a production to respiration ratio smaller than one and a dominance of collectors for large river sections. Junk et al. (1989) highlighted the importance of flood pulses in river-floodplain systems, whereas floodplain size generally increases with river size as well as flood duration and predictability which in turn emphasises the importance of floodplains in large rivers. Ward (1989) conceptualised on the processes of riverine ecosystems by the four-dimensional nature of lotic ecosystems where beside the spatial, the temporal dimension was taken into account too. The dynamic and comprehensive nature of a large river ecosystem and the fundamental role of



processes acting on habitats like sedimentology is highlighted by e.g. Moss (2008). Others underlined the importance of scales and their hierarchy (e.g. Frissell et al., 1986; Poff, 1997) and yet others focussed on the patterns and amplitude of processes (e.g. Lake, 2000; Poff & Ward, 1990; Townsend, 1989; Townsend et al., 1997). In common, those concepts underpin the relevance of diverse habitats and dynamics of the inherent processes or try to unify those aspects (Thorp et al., 2006).

However, processes initiating habitat quality and availability are inevitably linked to typology of large rivers whereas braided sections form distinctive different templets than meandering courses. From a biological aspect these types are well reflected by the prevailing organisms under near natural conditions and have to be considered in assessment approaches.

Geology, climate, and basin land cover have been often reported as primary drivers of stream and river ecosystems (Bhowmik et al., 1984; Resh et al., 1988). Under undisturbed conditions, river processes and dynamics vary over time, responding to seasonal, annual, and long-term changes in the three drivers. Water and sediment discharge regimes within the basin stream network provide the major mechanisms for the drivers to affect changes in the river dynamics. Within a basin, as rivers increase in size in the downstream direction, predictable gradients occur in the forces that shape the river, control the substrate, and provide organic material (Webster & Patten, 1979; Vannote et al., 1980).

In summary, in large rivers the stability as well as the disturbance amplitude trigger integral ecosystem functions, which determine where and when habitats are available. Thus, the basis for a sufficient understanding on the effects of stressors and their role in faunal changes strongly depends on the understanding of the ecosystem processes and their interlinkages which determine ecosystem functioning. However, the complex linkages between these ecosystem processes, directly affecting the habitat of biota and therefore the biological state, are not sufficiently quantified respectively proven especially in large rivers.



The history of drivers and pressures in European large rivers

The EU FP7-project REFORM 'Restoring rivers FOR effective catchment Management' (Grant No 282656) addressed hydro-morphological river modifications and their restoration (i.e. rehabilitation, mitigation). Within the D3.4 larger rivers were discussed as satellite topic. This chapter, based on Van Geest et al. (2015), syntheses the findings on the historical trajectory of driving forces, river regulation and rehabilitation of large European rivers. This syntheses includes a summary for the timeline of occurrence of drivers and pressures based on six case-study rivers.

Nowadays, almost all large rivers in Europe are heavily regulated, channelized and dammed, which often started centuries ago. Consequently channels incised, floodplains aggradated and hydroperiods have been modified to support hydropower generation, navigation and freshwater supply, and to protect the hinterland from flooding (Tockner et al., 2009). Due to this multiple stressor complex, the identification of the primal causes for degradation is complicated. As a consequence, former extensive aquatic/terrestrial transition zones lack most of their basic ecological functions.

Many large rivers in Europe are affected by the same drivers and associated pressures. However, little is known, on the chrono-sequence (timeline) in the occurrence of these drivers and pressures. Accordingly, this synthesis addresses the following questions:

- Are there any differences in the timeline of occurrence of drivers and associated pressures between rivers?
- What are commonalities, and what are differences?
- Are these differences related to climate regions in Europe?

A description of the evolving stressors over time may give insight into the causes for major transition points for species composition in large river ecosystems.

The case study rivers

In Van Geest et al. (2015), six case studies have been described that are spread across Europe. These case studies are representative of various European conditions with regard to climate, hydro-morphology and catchment size (Table 1). The case studies are situated in six countries. All these rivers can be characterised as large rivers (catchment area larger than 10,000 km²), although they differ in climatic zone, river length, catchment size, discharge, slope and river style. Large rivers can be considered as unique ecosystems and results are difficult to generalize. Still these case studies together give a good impression on the present regulation and rehabilitation of large rivers in Europe.



	Trent	Delta Rhine	Ро	Ebro	Vistula	Danube (Delta)
Country	GB	NL	IT	Е	Р	RO
Climate	Atl	Atl	Alp/Cont	Med	Cont	Cont
Length (km)	275	1,250	650	930	1,048	2,857
Catchment (km ²)	10,466	185,260	74,000	85,530	194,700	801,463
Discharge (m ³ /s)	28.5	2,200	1,540	462	1,046	6,500

Table 1: General characteristics of the six case studies described in this chapter (based on Van Geest et al., 2015)

Qualitative historical timeline of occurrence of drivers and pressures

Detailed information regarding the extent of drivers and pressures (i.e. exact determination of time and intensity of occurrence) are scarce. Thus, we will mostly discuss the time line in a qualitative manner.

Differences between rivers

Between the case studies, there are large differences in timing of these pressures. Along the river Po, deforestation already occurred during Roman Age, while for the other rivers this started in the Middle Age. Although data are scarce, this resulted in a strong increase of sediment load into rivers, which probably have resulted in large changes in hydro-morphological functioning of these rivers.

Large scale embankment of the river Rhine already started in the 14th century, while for the other rivers the majority of the embankments have been constructed much later, in the 19th (Trent, Vistula) or 20th century (Po, Ebro, Vistula). As a result, large parts of the active floodplains were permanently cut off from river flooding and have been converted to agriculture land, resulting in a large loss of wetlands along rivers.

Before the 19th century, small adjustments were already made to facilitate navigation in rivers. From the start of the Industrial Revolution (at the end of the 19th century) however, the conditions for navigation were strongly improved by channelization, bank protection and dredging of the main river channel. Commonly, this has resulted in a huge loss of highly dynamic pioneer habitats along rivers. The construction of large dams was mainly concentrated in the second half of the 20th century, thereby changing water levels, sediment budgets and impeding longitudinal connectivity for migrating fish.



Table 2: Timing of most dominant pressures for hydro-morphology in the catchments of the rivers Trent, Po, Ebro, Delta Rhine, Vistula and Danube (Delta). When no information is given, then this pressure is not considered as an important pressure.

River	Deforestation catchment	Construction embankments	Channelisation, bank protection, dredging	Large dams	Water dams
Trent	Middle Age?	19th - today	19th - today	19th - today	
Delta Rhine	Middle Age	14th - today	19th - today		
Po	Roman time	20th	19th - today	20th	19th - today
Ebro	Middle Age - 1950	> 1950	> 1950	> 1950	> 1950
Vistula	Middle Age?	1850 - today	14th - 18th		
Danube (Delta)	Middle Age?	20th	1880 - 1990	> 1950	

In the text below, the timeline of each of different drivers (and associated pressures) are discussed in more detail.

Flood protection and agriculture

The primary drivers were flood protection and agriculture for early regulation of all rivers (Table 1 & Table 2). For many rivers, these forms of river regulation started already centuries ago. For the river Po, the human-induced changes became already important during the Roman age. In this period, the dramatic increase in agricultural development and deforestation strongly increased the sediment load of the river and caused an extension of the Po delta along the Adriatic coastline. Deforestation also had a strong influence on the hydro-morphology of other rivers (e.g. Ebro).

Additionally, large parts of (formerly active) floodplains were embanked in all case studies, both for flood protection and agricultural use of the land. These results are in line with many other rivers in northern temperate regions. For North-America and Europe, it has been estimated that approximately 90% of the original floodplain of rivers has been permanently cut off from river flooding by the construction of embankments (Tockner & Stanford, 2002) resulting in a huge loss of low-dynamic habitats in floodplains, such as wetlands and hardwood forest.

The river regulations at around 1800 resulted in a river landscape in which the floodplain area was strongly reduced, but with the possibility for the main channel to migrate freely in the remaining active floodplain, resulting in a dynamic landscape with regular rejuvenation.

Navigation

Especially from the start of the industrial revolution (at about 1850), there was a strong need to improve conditions for navigation. Channelization of rivers for shipping activities has a negative impact on the occurrence of highly dynamic habitats as a result of the stabilisation of the river bed (by groynes, bank protection) and by deepening of the main channel. From this moment onwards,



the position of the main channel became fixed and the area of shallow water in the main channel declined strongly.

Navigation plays an important role in many rivers in Europe, and accordingly, also for the case studies in this report. Of our case studies, only the river Vistula in Poland has not been regulated for navigation purposes, and – hence – large parts of the main channel of the river have not been channelized. Beside the hydro-morphological consequences of waterway constructions, navigation directly affects aquatic organisms by vessel-induced waves.

Construction of dams

In the decades after the World War II, many dams were constructed in the rivers. These dams were used for the generation of hydropower, as well as for water supply and irrigation. This has resulted in a decreased longitudinal and lateral connectivity, thereby impeding conditions for migratory fish and other species. Additionally, the construction of the dams resulted in altered hydrological regimes in rivers and reduced the sediment supply to downstream sections, as well to river deltas. Of the case studies, especially the rivers Trent, Po, Ebro and Lower Danube have been severely impacted by the construction of dams.

Pressure effects on hydromorphological processes and ecology

For the majority of the case studies, only limited information was available regarding the observed impacts of pressures on hydro-morphology and ecology. It seems that the sequence of drivers (and associated pressures, see above) have initiated major transition points for ecological processes and biota along large rivers. In the following, the main results are discussed in respect to the time line for the occurrence of these drivers and pressures.

Effects of deforestation

Many catchments of rivers have been strongly affected by deforestation. This has already started many centuries ago, and occurred during the Roman age (river Po) to medieval times (for many other rivers). As a result of deforestation, there was a strong increase in runoff of sediments into the river, which has a strong impact on the hydro-morphological processes and - hence – on river style. Although this must have had a large impact on river systems, there is a large lack of knowledge on the effects on hydro-morphological processes, as well on ecological processes and species composition. Moreover, gradual changes in climate (e.g. increased precipitation) may have caused similar changes to river systems, by changing vegetation composition and runoff patterns of river catchments.

Effects of embankments

The construction of embankments has resulted in a strong reduction of the active floodplains along rivers. Along the Delta Rhine, the river was already completely embanked at about 1400, but for other case studies this started from the 19th century (Table 2). Almost all rivers in Europe are embanked, and hence there are only a few examples left of extensive, intact floodplains. The case



study of the Danube Delta may serve as an example for such systems. In extensive floodplains, there are clear gradients in hydrologic residence time i.e. water age and hydraulic resistance, resulting in gradients of sedimentation and nutrients along the lateral dimension of floodplains. Along embanked rivers, however, such gradients are strongly shortened, because of the reduced width of the active floodplain.

Effects of pressures related to navigation

In Europe, large parts of river floodplains were already embanked and floodplains were partly used for agricultural purposes. However, the main channel river was still able to change its course, resulting in a more or less continuous formation of new habitats, such as islands, point bars and abandoned channels. The main channel was still shallow, and in the river-bed there was a gradient from coarse sands in the erosive zones to silt in the depositional areas. Thus, although these rivers did not represent pristine conditions, they were still dynamic processes and according habitats with extensive land use that was largely adapted to the natural morphological patterns and processes (Middelkoop et al, 2005). Paleolimnological research along the Delta Rhine indicates that parts of the river banks may have been covered by macrophytes, while dead trees provided snag habitats which were important for a large number of macroinvertebrate species (e.g. Simulidae; Klink, 1989).

For many rivers, at that time there have already been minor adjustments to the river bed to facilitate navigation. From the start of the Industrial Revolution however, there was a strong need to improve conditions for navigation. Consequently, groynes were constructed, river banks of the main channel were protected with rip-rap and the main channel was deepened. Additionally, dead wood (snag habitat) was removed. As a result, the rejuvenation of the landscape stopped due to the fixed position of the main channel. This has had strong impact on hydro-morphological processes, habitats and thus species composition.

Because of the fixed main channel, the continuous formation of new habitats ceased while succession continued, and thus the overall landscape age increases. This has resulted in a dramatic change in riverine landscape composition in favour of species typical for less dynamic habitats, as was shown and discussed in several studies (Petts & Amoros, 1996; Johnson, 1997; Hughes, 2001; Marston et al., 1995; Tockner & Stanford, 2002). Those changes are strongly related to a legacy effect, as it takes quite some time before their effects get visible. An observed high biodiversity is often a relict of former conditions that will develop towards a lower diversity and a shift in landscape position (Geerling, 2008; Bravard et al., 1986; Tockner and Stanford, 2002). In such a setting, floodplain age distribution can develop as shown in Fig. 2.

In addition to rejuvenation, succession will also cease as land use in the floodplains (partially) changes towards agriculture, as is the case along many regulated rivers. Some ecotopes are converted to pastures or fields, while other will remain in a more natural state. The latter will become relic ecotopes that stay in ecological succession, e.g. relic disconnected side channels. Such a landscape has 'gaps' in its age distribution; i.e. it is a temporal discontinuous landscape



(Geerling, 2008; Fig. 3). Although data is scarce, it can be assumed that this has resulted in a strong decline of many riverine species. Nowadays, a large number of riverine species characteristic for young, dynamic habitats are extinct or have strongly declined in number.

Construction of dams

In many rivers, large dams have been constructed, especially after the World War II. There are many comprehensive reviews of the hydro-morphological effects and ecological impacts downstream of dams (e.g. García de Jálon et al., 2013 and references in this report). Overall, the ecological impact of dams often result in three types of environmental alterations (Rood, 2005): (1) changes in the released flow regime (quantity and quality); (2) reduced passage of alluvial materials, in particular suspended solids, and (3) fragmentation of the river corridor, resulting in interruptions in downstream and upstream passage of biota (e.g. fish species).



Age of ecotopes

Fig. 2 Conceptual graph of a hypothetical area versus age of natural ecotopes of river with habitat rejuvenation (solid line) versus a regulated river without rejuvenation (dotted line). Along regulated rivers, existing ecotopes continue their succession, while pioneer sites are disappearing (after Geerling, 2008).



Age of ecotopes





Conclusions on the stressor history in the case studies

Humans have a long history for affecting the hydro-morphological and ecological functioning of rivers. Along the river Po, deforestation started already during Roman Age, while for the other rivers this occurred in the Middle Age. In subsequent centuries, the rivers were embanked (Rhine: 14th century; other rivers 19th and 20th century) and the position of the main channel was fixed with groynes, bank protection and dredging (19th/20th century). After 1950, many dams were constructed.

Although data are scarce, several major transition points can be expected during these developments. The first transition point probably took place during deforestation of the catchment, which strongly affected hydro-morphological functioning of rivers. Another major transition point for river ecosystems is the large scale embankment of lowland rivers, which resulted in a large loss of wetland habitats and hardwood forest. However, at that time there was still the possibility for the main channel to migrate freely in the remaining active floodplain, resulting in a dynamic landscape with regular rejuvenation. Nowadays, these habitats have disappeared due to the fixation of the main channel with groynes, bank protection and dredging. As a result of all these modifications, the former extensive aquatic/terrestrial zones along large rivers in Europe lack most of their basic ecological functions, and the habitats for their characteristic species composition have strongly deteriorated.



Selected quantitative stressor timelines in large European rivers

The summary of the REFORM case studies in the preceding chapter gives a good overview of the general temporal development of human-induced changes in large European rivers. Due to the fact that major parts of the Europe consist of cultural landscape, large rivers in Central Europe have been tremendously altered by diverse human impacts over the last centuries (Jungwirth et al., 2014; Petts et al., 1989). After the Industrial Revolution starting from the end of the 19th century, the pronounced channelization of rivers, the construction hydropower plants and damming led to a completely different hydro-morphology of the rivers with a strong decrease in typical dynamic processes. Different stressors interfere with each other and create a multi-stressor complex (Fig. 4). This chapter focuses on selected stressor developments in the 20th century.



Fig. 4: Conceptual description of multi-stressor development and overlay over time

Information on the temporal development of stressors was derived by reviewing a magnitude of sources. Beside scientific literature and books, we included data from statistical services (http://ec.europa.eu/eurostat/, http://www.statistik.at/), internet resources, e.g. webpages of hydropower companies, localities, and grey literature, which provided information on the historical development of human-induced changes of large European rivers.

Hydromorphological changes

Hydromorphology, as defined by the Water Framework Directive (WFD), represents the physical characteristics of the shape, boundaries and content of a water body. Hence, the term hydromorphological changes summarises several effects on riverine ecosystems which in turn impact different ecosystem processes. In the following the development and effects of channelization and damming are discussed in more detail.



Channelization

Large rivers and – connected with them – floodplains are sensitive and complex ecosystems which are strongly determined by hydrological processes. Lateral connectivity and interactions between river and floodplain are essential processes for the ecosystem-functioning (Amoros & Roux, 1988; Gergel et al., 2002; Junk et al., 1989; Schiemer and Zalewski, 1992; Ward, 1989).

A precise summary of the well documented development and progressing channelization at the Danube is given by Demek et al. (2008). The first systematic large-scale channelization schemes at the Upper Danube River and the Upper Rhine River were initiated as early as the end of the Napoleonic Wars (1805 – 1815; Pasetti, 1862). Hohensinner (2008) and Hohensinner et al. (2004, 2005, 2008a, b) describe in detail the development of channelization at the Austrian Danube since the early 18th century. In Fig. 5 (right) the hydromorphological changes from 1715 up to now are illustrated. Based on the digitisation of different habitat types sensu Amoros et al. (1987) the hydromorphological change over time can be described by the change of the habitats (Fig. 6). All habitats have been reduced in size. However, after all channelization works more or less only main channel habitat types is schematically related with the turn-over of different functional groups and the loss of biodiversity.



Fig. 5 Terrestrialisation processes due to river regulation and faunal reaction (left: Ward et al. 2002; right: Graf et al. 2013, Danube river at Vienna)







Fig. 6: Temporal change of habitat types from 1812 to 2006 on a Danube River stretch in Austria (Hohensinner et al., 2011), AZ=active zone, intens. = intensive channelization

Damming

In general, damming leads to increasing sedimentation of fine particles due to the reduction of current velocity in longitudinal, lateral and vertical (clogging of the interstitial) dimensions (Moog, 1986; Banning, 1998). Faunal changes are well documented and have different extend from headrace to the weir (Herzig et al., 1987; Moog & Jungwirth, 1992). In general a dramatic change of functional groups from rheophilous to stagnophilous organisms and from scraper/filter feeders to detritivorous species can be observed. Due to enhanced autotrophic production in dammed areas the nutrient cycle is altered and filter-feeding assemblages increase below dams (e.g. Statzner, 1981; Mauch, 1981). Besides these local impacts, damming influences the discharge regime and sediment transport considerably and changes the overall character of riverine systems (e.g. Habersack et al., 2013). The homogenized discharge-dynamics and summation effects of dam-chains lead to a loss of type-specific organisms, which are replaced by pioneers and more opportunistic and insensitive faunal elements (Resh et al., 1988).



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 7: Number of dams along Danube, Rhine and Weser since 1860

River bed incision is a direct consequence of channelization and sediment retention due to damming. The reduced flow velocities upstream of a dam lead to sediment deposition. Downstream of the dam, the river tries to compensate for this deficit in bed load material and the river bottom is more intensely scoured followed by a drop of the water table. This water table drop also leads to a reduced lateral connectivity between the main river and the floodplains. This degradation process can be observed on several river stretches across Europe. Fig. 8 shows the progressing incision of the Danube River between Vienna and Bratislava between 1950 and 2003. In 1991, the construction of the hydropower plant Freudenau was finished followed by a distinctive drop of the water table.



Fig. 8. The low water table of the Danube from 1950 to 2003 (gauge at Wildungsmauer, river-km 1894.7); m.a.sl. - metres above sea level. (Reckendorfer et al., 2005)



Pollution

Water quality pollution is a major stressor of rivers on a European scale. The water of rivers represented not only one of the most important large distance transportation system for societies in earlier days, but also the most important mean of waste transport. Both are fundamental reasons why large settlements and cities preferably developed next to rivers and in a lot of cases next to large rivers.

An excellent description of various pollution pathways in Vienna during the Middle Ages is given by Kohl (2010), which may be generally applied on most European cities and connected large rivers of that time. Liebmann & Reichenbach-Klinke (1967) list pollution sources along the entire course of the Danube and provide a historical outline of organic pollution (e.g. the first biological water quality map of the Austrian Danube). As one example of large rivers, Tobias (1996) gives an overview of the development of the oxygen- and ammonium content from 1970 to 1994 at the river Main with highest pollution loads between 1972 and 1980, and a recovery afterwards which clearly correlates with the revival of the mayfly *Ephoron virgo*. Since that time water quality has substantially enhanced during the last decades, mainly because of raised environmental awareness based on continuous saprobiological surveys and subsequent improved purification processes.

Organic pollution has generally lost its primary role as stressor in aquatic systems of Central Europe and has been replaced nowadays by hydromorphological degradation. Anyhow, organic pollution had its negative effects in the past, and detailed monitoring campaigns have impressively initiated a reduction of organic pollution in the Danube (e.g. Jungwirth et al., 2014). In regard of water chemistry, hazardous and endocrine substances, which impact biological quality elements, are a currently emerging issue in water management. The effects of today's applied substances in agriculture as well as in industrial processes together with effluents of sewage treatment plants and their combined effects via the whole catchment areas are poorly understood.



Navigation & Neozoa

The following chapter is dedicated to two special components of human-used large river systems whose occurrences and intensities are interlinked. Navigation on rivers is accompanied by several impacts on the ecosystem, which directly and indirectly affect the aquatic fauna. Likewise, the distribution of neozoa is also triggered by navigation. Accordingly, both stressors are commonly discussed.

Navigation

Most of the dams shown in Fig. 7 were built for power generation but on the River Weser the dams were actually built to provide constant water depth for vessels. Also on the Danube, the construction of dams led to an improved navigability of shallow passages as well as sections with turbulent currents. Accompanying, navigation channels were constructed to improve the waterways with according effects on river morphology.

In 1954 the Conférence Européenne des Ministres de Transports (CEMT) established an international classification system dividing the European waterways into five classes, depending on their dimensions based on the dimensions of five vessel types. Accordingly, a waterway's class was determined by the largest standard vessel it can accommodate at which the width was the main determining factor. The first convoy of push barges travelled along the Rhine in 1957 with a subsequent success for inland water transport. The CEMT responded in 1961 by adding Class VI to its classification, which became inadequate too soon. In a new, uniform classification drawn up by the CEMT and the UN's Economic Commission for Europe (ECE), known as CEMT1992 (Fig. 9), on the one hand East European waterways were taken into account and on the other hand larger size classes were included.

	Type de voies navigables Type of Inland	Type de voies navigables Type of Inland	Type de voies navigables Type of Inland	Type de voies navigables Type of Inland	Type de voies navigables	Classe de voies navigables		Automot Motor ve	eurs et chalan ssels and barg	ds ges			Co Pu:	nvois pous shed convo	isés Dys		Hauteur minimale sous les ponts
					of navigables waterways	Ty Ty	pe de bateau pe of vessel: ¿	<: caractéristiq générales char	ues générales acteristics		Type de co Type of co	onvoi- Ca onvoy- Gé	ractéristiqu nérales cha	ies générales aracteristics		Minimum height under	
	waterways		Dénomination Designation	Longueur Length	Largeur Beam	Tirant déau Draught	Tonnage Tonnage		Longueur Length	Largeur Beam	Tirant déau Draught	Tonnage Tonnage	bridges				
				m	m	m	t		m	m	m	t	m				
VTEREI	OF RE	I	Péniche Barge	38.50	5.05	1.80-2.20	250-400						4.00				
- REGIO	GION, RTANO	Ш	Kast-Caminois Campine-Barge	50 - 55	6.60	2.50	4.00-650						4.00-5.00				
DNAL	₩₽	Ш	Gustav Koenings	67 - 80	8.20	2.50	650-1000						4.00-5.00				
Г		IV	Johan Welker	80-85	9.50	2.50	1000-1500	-	85	9.50	2.50-2.80	1250 - 1450	5.25/or 7.00				
D'I		Va	Grand bateaux Rhenands/Large Rhine Vessels	95-110	11.40	2.50-2.80	1500-3000		95-110	11.40	2.50-4.50	1600- 3000	5.25/or 7.00/or				
VTERE		Vb						-	172-185	11.40	2.50-4.50	3200- 6000	9.10				
TINTE	ITERN,	Vla						-	95-110	22.80	2.50-4.50	3200- 6000	7.10/or 9.10				
RNATIONAL	ATIO	VIb		140	15.00	3.90		-	185-195	22.80	2,50-4,50	6400- 12000	7.10/or 9.10				
	: NAL	VIc							270-280 193-200	22.80 33.00-34.20	2.50-4.50 2.50-4.50	9600- 18000	9.10				
		VII							285 195	33.00 34.20	2.50-4.50	14500- 27000	9.10				

Fig. 9: CEMT classification of waterways from 1992 (source: http://www.pianc.org/)

After the success of push barges, container ships are increasingly in use on inland waterways since the late 1970s as inland navigation vessels are perfect for transporting containers with their rectangular holds.

Table 3: Properties of container transport on the Rhine in the year 1996, 2001, and 2006; TEU = Twenty-foot Equivalent Unit, which characterises the capacity of vessels (source: http://www.watererfgoed.nl/)

property	1996	2001	2006
average carrying capacity (survey data)	185 TEU	273 TEU	310 TEU
average number of containers (survey)	122 TEU	165 TEU	188 TEU
average load factor	66%	61%	61%
proportion of containers loaded	73%	67%	70%
- upriver	59%	49%	56%
- downriver	88%	86%	87%
average weight of all loaded containers*	12.4 t/TEU	12.7 t/TEU	11.4 t/TEU
- upriver*	12.1 t/TEU	11.6 t/TEU	10.6 t/TEU
- downriver*	13.7 t/TEU	13.8 t/TEU	12.4 t/TEU
proportion of 20-foot containers	55%	31%	32%
proportion of 40-foot containers	45%	68%	66%
proportion of 45-foot containers	0%	1%	2%
proportion of high cube containers	7%	18%	31%
proportion of pallet wide containers	0%	0%	3%

* incl. weight of empty container; percentages refer to proportion of TEU

The temporal comparison of different features of container transport on the Rhine, shown in Table 3, reveals that the capacity of container ships is growing steadily. The proportion of 40-foot containers rose from 45% to 66%, and the proportion of high cubes from 7% to 31% within the ten years from 1996 to 2006.

In summary, this underlines the increasing size of vessels which are used for inland navigation and therefore the increasing navigation intensity. The amount of transported goods doubled on the Austrian Danube with a sharp increase after the collapse of the Eastern Bloc at the beginning of the 1990s (Fig. 10). Only the amount of inland transport stagnated.

Deliverable 5.C Report on legacy and tipping points in large rivers





Fig. 10: Transported goods in million tons on the Danube River in Austria from 1959 to 2014

14

12

10

Total -Import

> Transit Export

Inland

This trend is totally underpinned by Fig. 11, which reveals a continuous and sharp increase of average vessel size between 1970 and 2010, with almost tripling the average vessel size.



Fig. 11: Temporal development of average vessel size (indicated by carrying capacity) from 1970 to 2010 on the Scheldt-Rhine Canal (Volkerak Locks), the Lek Canal (Pr. Beatrix Locks & Pr. Margriet Locks), and the Rhine (Lobith) (source: http://www.watererfgoed.nl/)

Beside the morphological changes as accompaniment of navigation on rivers, vessel induced waves lead to high shear stress at the river banks (Liedermann et al., 2014). Liebmann & Reichenbach-Klinke (1967) already observed severe negative effects by navigation, especially caused by wave action. Juvenile fish were reported to be hurled at the riparian zone, fish were disturbed during spawning in general, and oil etc. was polluting the substrate. Especially wave wash effects have impacts on juvenile fish as reported by Hirzinger (2002), Kucera-Hirzinger



(2009) and Schludermann et al. (2014). Gabel et al. (2008; 2011a, 2011b) investigated the reactions of selected macroinvertebrates and their interactions with fish under the influence of wave actions.

Negative effects on merolimnic organisms by mechanical damaging, especially during moulting processes at the shoreline can be expected but have not studied yet in detail; in fact the majority of insects still persisting nowadays in the Danube moult nearly exclusively at the water surface. Furthermore ships are generally suggested to enhance the spreading of neozoa as vectors through ballast water and vessel hulls as suitable colonising substrate.

<u>Neozoa</u>

Neozoa are per definition species, which colonise a given area after the year 1492. Reliable studies on macroinvertebrates started with Linnaeus back at the end of the 18th century, which made the designation of certain species difficult due to lack of detailed distributional information. Current and historical zoogeographical patterns are mainly the result of climatic conditions. Accordingly, various shifts, either recent or historic, have been documented. For example *Dreissena polymorpha* is documented from tertiarian times in Central Europe, survived glaciation in southern areas and returned during the 18th century (Grossinger 1794). Species ranges have been and will be oscillating, but anthropogenic induced pressures, like navigation or climate change, speed up these processes.

The increasing occurrence of invasive alien species in connection with the increasingly documented loss of indigenous faunas of large rivers is observed on a European-wide scale (e.g., Moog et al., 2007; Graf et al., 2008). Besides biodiversity issues, this phenomenon is intensively discussed in the context of ecological assessment systems and the closely linked management actions (Orendt et al., 2009).

The Danube River is - besides a northern corridor via the Volga to the Baltic Sea, and a central pathway via the Dnieper to the Elbe and the Rhine – the main southern migration route of aquatic Ponto-Caspian elements (Bij de Vaate et al., 2002) and the majority of neozoa in the Danube therefore clearly belong to Crustacea and Mollusca from this region. Fig. 12 describes the distribution of the genera Amphipoda with densities along the Danube River. Only the genus Gammarus is considered to be native in the Upper and Middle Danube. Accordingly, the whole Danube River is nowadays home of the different genera of Amphipoda.





Fig. 12: Distribution of Amphipoda-genera with densities along the Danube based on JDS 2 data

Direct negative influences of invasive alien species on the original fauna have been hardly testified, but Schöll (2006) found clear correlations between increasing densities of the amphipod *Dikerogammarus villosus* and the population-decrease of the caddisfly genus Hydropsyche in the Rhine River. Moog et al. (2013) described similar interactions between *D. villosus* and *Gammarus fossarum* and *G. roeseli* respectively in the river Traun. According to Pöckl (2006) the predator *D. villosus* showed higher fertility than the resident *G. fossarum* and *G. roeselii* and is successfully competing with them. Bacela et al. (2008) also stated significant changes among the benthic associations after the new colonisation of *D. villosus* in Rhine, Oder, Danube and Meuse. Nowak (2012) investigated the effect of *Dreissena bugensis* on other benthic invertebrates, but the processes behind species competition are still poorly understood.

The seriousness of this problem may be illustrated exemplarily by the recently documented structure of benthic assemblages of the Danube River during the JDS 2 (ICPDR 2008) expeditions which sampled along the whole river: Among the ten most frequent macroinvertebrate species sampled, nine are assigned as neozoa (Graf et al., 2008), above all occurring in very high densities and frequency (Fig. 13).



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 13: Occurrence, frequency and abundance of Peracarida in the Danube River. *: Absent in the Lower Danube; **: Occurrence also in the Middle Danube, downstream of Belgrade. In terms of abundance neozoa dominate clearly the benthic communities and reached up to 50% of all documented taxa in the Upper and Middle Danube

Neozoa are characterised by Statzner et al. (2008) as ecological flexible, having high fertility rates and as non-sensitive, thereby being more robust, which enables them to colonise impacted environments. In fact, large river ecosystems are multiple-stressed and invasive species may just fill up empty niches after the loss of indigenous elements or outcompete the impaired populations. Analysing the enhanced invasions in Austria since the 1980s, Moog (2010) and Korte & Sommerhäuser (2012) mention the increasing water temperatures as one essential trigger, which was revealed earlier by Rahel & Olden (2008) too. Another important driver, which represents a pressure to other faunal elements too, is navigation. Beside the direct transport of individual through ballast water or on the outside of the vessel, the hydromorphological changes which are implemented on waterways favour the establishment of neozoa.

From an ecological point, the most dominant neozoa have severe impacts on the entire functioning of aquatic ecosystems as they (1) reach high densities (e.g. 500,000 ind./ m² of *Chelicocorophium curvispinum* in the Morava (Graf et al., 2005), dominate the benthic community and colonise niches of indigenous faunas, (2) act partly as bio-engineers changing the habitat characteristics entirely (*Chelicocorophium* spp. alters the microhabitat structures by building tubes, *Corbicula* spp. provides a specific habitat for other species respectively as the diameter of adult shells resembles microlithal conditions), and (3) intervene significantly in the nutrient cycle e.g. Corbicula spp. This Asian clam – an active filter feeder – shows mass occurrence and can reach a biomass of more than 7 kg/m² (Moog et al., 2007; Danube at Linz, Austria).

Linking shipping intensity (indicated by the amount of transported goods) and the number of neozoa foung along the Austrian Danube shows a clear coinciding pattern. Again, the socio-political event of the fall of the Eastern Bloc is remarkably recognisable (Fig. 14).



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 14: Development of total tonnage freight and number of neozoa in the Austrian section of the Danube River from 1959 to 2013 (data sources: Statistik Austria, Moog et al., 2013)



Large river fauna

Beside their hydro-morphological characteristics, large rivers also feature and especially featured typical faunal elements and species. However, referring to historical communities and species, which a river once featured, is possible for economically relevant fish species like sturgeons but difficult for other biological elements like Plecoptera (Tittizer et al., 1991).

The analyses of faunal changes address two aspects: (1) disappearance of typical Plecoptera species in large European rivers and the change of distribution patterns over time, and (2) the changes and shifts in fish assemblages of large rivers over time based on the example of the Danube River.

Changes in the Plecoptera fauna of large European rivers

Many of the species occurring in large rivers typically covered large areas in Europe (summarised exemplarily for Plecoptera by Zwick, 1992 and Graf et al., 2008). Accordingly, we can assume that the Plecoptera fauna of large rivers has been uniformly shaped with slight replacement of sibling species at zoogeographical borders. These 'legends of large rivers' still persist in discrete refugia. Hence, they are able to indicate sites with ecologically integer processes, potentially serving as reference to identify those ecosystem processes which are necessary to sustain faunal elements typical for large rivers (e.g. Bojková, 2009; Bojková et al., 2011; 2012a; 2012b; 2013; Graf & Kovács, 2002; Kovács et al., 2004; Ruffoni & Le Doaré, 2009; Greulich, 2014; Chovet & Lecureuil, 2009; Wantzen & Richard, 2014) scattered across Europe.

Definition of species of interest

The species records based on spatial explicit occurrences is highly important to link species occurrences with stressors through space and time. Principally, two major criteria were applied to identify Plecoptera species of interest for the purposes of this report. Firstly, the species have a high conservation status mirrored by the Red List classification, i.e. critically endangered or extinct. Secondly, the species were once wide-spread and potamotypic (large river) species. This criterion was checked by information from http://www.freshwaterecology.info (Graf et al., 2008).

The International Union for Conservation of Nature (IUCN) Red List of Threatened Species (further called 'Red List') is the most comprehensive inventory of the conservation status of biological species. The conservation status is identified on the basis of assessments on regular intervals and provides information on range, population size, habitat and ecology, use and trade, as well as threats of species. Based on these criteria, the Red Lists classify species into different threatening categories (Fig. 15). Basically, the feasibility of this approach differs between taxonomic groups, especially on the global scale. Hence, Regional Red Lists report the threatening status of species within a certain country or management unit. Therefore giving the possibility to (1) implement the approach for species which are not of global concern, and (2) to directly feed into management actions. The creation of a Regional Red List has to follow a clear and repeatable protocol.



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 15: IUCN Red List categories, EX=extinct, EW=extinct in the wild, CR=critically endangered, EN=endangered, VU=vulnerable, NT=near threatened, LC=least concern; not shown: DD=data deficient, NE=not evaluated (source: www.iucn.org)

As an initial step we collected and screened several Regional Red Lists to identify, underline and highlight the conservation status and extinction risk of Plecoptera typically occurring in large European rivers.

A Red List for Plecoptera was available for the following regions: Switzerland (Lubini et al. 2012), Germany (Reusch & Weinzierl, 2001; Saxony Voigt & Küttner, 2015; Brandenburg (Braasch & Berger, 2003), Bavaria (Weinzierl, 2003), Hesse (Wolf & Widdig, 2013), Sachsen-Anhalt (Böhme, 2004)), Czech Republic (Bojková & Soldán, 2013), Poland (Fialkowski & Sowa, 2002) Norway (Kjærstad et al., 2010), Serbia (Petrović et al., 2014), Spain (Verdú & Galante, 2006), Croatia (Popijac, 2008), Slovenia (Sivec, 2002), Great Britain (Craig, 2015). Additionally comments on the status of species was included from the following sources: Boumans (2011), Bojková et al. (2012a), Bojková et al. (2012b), Popijac & Sivec (2009);

Based on the criteria the following species were defined as the species of interest:

```
Chloroperlidae
```

Xanthoperla apicalis (Newman, 1836)

```
Perlidae
```

Agnetina elegantula (Klapálek, 1905)

```
Marthamea vitripennis (Burmeister, 1839)
```

```
Perlodidae
```

Besdolus ventralis (Pictet, 1841) Isogenus nubecula (Newman, 1833) Isoperla obscura (Zetterstedt, 1840) Isoperla pawlowskii (Wojtas, 1961) Perlodes dispar (Rambur, 1842) Taeniopterygidae Brachyptera braueri (Klapálek, 1900)



Brachyptera trifasciata (Pictet, 1832)

Sources for species records of Plecoptera

Initially, stonefly data were extracted from the so-called DAEP database (Schmidt-Kloiber et al., 2016; Graf et al., 2016). The 'Distribution Atlas of European Plecoptera' (DAEP) database is an initiative which have been started within the EU-funded project BioFresh (Contract No. 226874, www.freshwaterbiodiversity.eu). DAEP aims at compiling as many (adult) stonefly data from Europe as possible to increase the knowledge about distributions over space and time of this highly sensitive organism group. Even after the end of the BioFresh project in the year 2014, the work on the database was continued, mostly focusing on filling the distribution gaps and an extensive quality control.

The database currently holds about 100,000 occurrence records of stoneflies from all over Europe. Occurrence data in the dataset were provided by 20 data holders. All data will be made available through the Freshwater Biodiversity Data Portal (www.freshwaterplatform.eu) as soon as data compilation and quality control is finished.

Based on the information from DAEP, we extended the species records by screening taxonomical papers and records described in other literature sources (e.g. Bojková, 2009; Bojková et al., 2011, 2012a; 2012b; 2013; Graf & Kovács, 2002; Kovács et al., 2004; Ruffoni & Le Doaré, 2009). The combination of the data led to a total of 1742 species records across Europe (Fig. 16).

Spatial and temporal units of analyses

The analyses to investigate changes in the distribution of Plecoptera cover a spatial and a temporal level. The temporal level is principally given by the date of the occurrence record. For further analyses, the dataset was split into two parts comparing the periods before and after 1990. The year 1990 was chosen because at this time the human-induced changes such as river straightening or damming (e.g. the construction of the last hydropower plant on the Danube in Austria started in 1992) as well as the impact of those changes on the ecosystems have been in full operation.

The spatial level was handled by two approaches: firstly, the species records were analysed on the bases of their location, i.e. coordinates. In the second approach, we delineated in total 92 large river catchments across Europe based on the CCM v2.1 (Vogt et al., 2007). In some cases, these catchments summarise areas without a large river (e.g. southern Scandinavia). Based on the catchments, the records were intersected with those spatial units. After the intersection, only the information, if a species occurred or not was kept. By this procedure we accounted for inhomogeneous sampling efforts especially for accumulated records within small areas (e.g. Raba catchment). Furthermore, we assumed that if a species occurred in the period after 1990 that this species must have been present in the period prior to 1990, too.



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 16: Location of the species records for the ten Plecoptera species of interest (N=1742)

Results & Discussion

Large rivers in Europe have undergone many anthropogenic modifications and have lost a high share of their indigenous fauna, especially sensitive insects like Plecopterans. Nowadays, a majority of them is found on Red Lists of different countries as threatened or even extinct, which represent a high threatening status (see Fig. 15). Many of the once abundant and characteristic species of large rivers have drastically declined in their abundance or are even extinct in Central Europe, since their habitats have either completely disappeared or are too fragmented for natural recruitment in large abundances.

The spatial distribution of the records (Fig. 18) and large river catchments with species records of the Plecoptera species of interest before and after 1990 (Fig. 19) showed a reduced distribution across Europe. Species records especially reduced in the northern and western part of Europe. Interestingly, the total number of records does not differ that much (943 before/799 after 1990). Comparing the number of records per catchment (Fig. 17), it gets obvious that especially the number of catchments with 20 to 50 records decreased. Species-specific maps showing the record locations as well as the large river catchments with records for both periods can be found in the appendix (Fig. 29 to Fig. 38).



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 17: Number of species records per catchment before 1990 (left) and after 1990 (right)



Fig. 18: Distribution of record points of all ten Plecoptera species of interest before 1990 (left) and after 1990 (right)



Fig. 19: Distribution of large river catchments with records of all ten Plecoptera species of interest before 1990 (left) and after 1990 (right)





Based on the intersection between large river catchments and record locations it was possible to identify the number of catchments with species records before and after 1990 (Fig. 20). Each species (with exception of *A. elegantula*) showed a reduced number of catchments with records for the period after 1990. The strongest decrease in the number of catchments with records occurred for *I. obscura*. In total, less than the half number of catchments were found with a species record after 1990. *A. elegantula*, *B. ventralis*, and *I. pawlowskii* revealed for both periods a relatively small number of records.



Fig. 20: Number of large river catchments with occurrence records of the ten Plecoptera species of interest and in total of all species before 1990 (turquoise) and after 1990 (purple)

Summing up the occurrence of the ten single species per period and catchment underlines the shrinkage in the distribution of those large river specialists (Fig. 21). Before 1990 eleven large river catchments featured more than five of the Plecoptera species of interest. After 1990 only two areas remained with five or more of those species, namely the Raba catchment and the middle section of the Danube River including the rivers Vah, Hron and Ipel. In turn, no records were identified for a large number of areas formerly featuring large river Plecoptera. Furthermore, some catchment areas kept a stable number of species from the period before to the period after 1990. These catchments include the following river basins: Loire incl. Allier, Drava, Inn, Nemunas and the upper part of the Tisza Basin.



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 21: Number of recorded species per catchment before 1990 (left) and after 1990 (middle) as well as summarising the number of catchments per record number category (right)

The coordinates of the record points were used to compare the distribution ranges of each species as well as the distribution centres (indicated by the median) between the two periods (Fig. 23). Interestingly, *A. elegantula*, which showed a larger number of catchments with records after 1990, revealed a shrinkage in latitude as well as a shift in both latitude and longitude. The records of *B. ventralis* shifted eastwards for the period after 1990. The range of distributions shrank in latitude and longitude for *B. braueri*. *B. trifasciata* shifted eastwards after 1990. Even though the number of catchments clearly decreased for *I. nubecula*, the species showed no shrinkage in longitudes. In turn the latitudes indicated a distinct shrinkage. *I. obscura*, which showed the largest decrease in the number of catchments with records, revealed no shrinkage neither for latitude nor for longitude. However, the median of latitude clearly shifted southwards. The records of *M. vitripennis* clearly shrank for both latitude and longitude. The latitudes of *P. dispar* records shrank and shifted southwards for the period after 1990 with comparable ranges of longitudes for both periods. *X. apicalis* showed the largest shrinkage in latitude as well as a westward shift of longitudes.





Fig. 22: Shift in latitude (left) and longitude (right) for species records of the ten Plecoptera species before 1990 (grey boxes) and after 1990 (white boxes)

There are some examples from literature on typical faunal elements of large rivers in Europe, which indicate that these species once occurred in very large numbers – in such large numbers that the presence of the species was not only recognised by taxonomical specialists or ecologists but by ordinary people too. Bridges in Prague were so crowded with the nowadays nearly vanished B. braueri that the public called it the "Prague fly". Calderini (1868) described the disturbance of local people by masses of B. trifasciata in Italy, and Ausserer (1869) mentioned this species to be "specialmente in primavera molto comune in tutta la fauna", i.e. the species have been very common for the fauna during spring. Kühtreiber (1934) remarked "all silts and sand banks are teeming with them", giving in turn a hint on the preferred substrate type by this species. I. nubecula was described in Brauer & Löw (1857) as "very common" for the Danube in the vicinity of Vienna. Mass emergence of the species Oemopteryx loewii was reported as early as 1775 by Schäffer from Regensburg, of which nowadays only few females are left in museums. The last reliable finding is reported by Russev (1962) from the Bulgarian Danube in 1955. In the Danube, Plecopterans could not be found downstream of site Oberloiben (Joint Danube Survey 2; ICPDR, 2008) while Raušer (1957) reported a rich indigenous stonefly community for the Danube and listed the following well documented species: B. trifasciata, B. braueri, O. loewii, Taeniopteryx araneoides, T. nebulosa, P. dispar, I. nubecula, I. obscura, I. difformis, M. vitripennis, X. apicalis and *Isoptena serricornis*. Other examples which demonstrate similar fates of large river species are given in e.g. Fittkau & Reiss (1983), Zwick (1984; 1992) and Fochetti & Tierno de Figueroa (2006).

Although cumulative effects of multiple stressor interactions are responsible for these losses, the last records of conspicuous species are well coinciding with the period of dam-building at the upper Danube. Zwick (1992) cites records of *I. nubecula* from England, France (Paris), the Netherlands, for the Danube at Ulm and Vienna, Dresden, and Bulgaria, and similar large areas have been covered once by *M. vitripennis* (Zwick 1984) and *X. apicalis* (Zwick 1999). Today's



populations are isolated and persisted exclusively in small and severely fragmented refuges as in the case of *I. nubecula* in the river system Lafnitz/Raba in Austria/Hungary and the Tisza in Hungary (Graf & Kovács, 2002; Kovács et al., 2002).

A few of these species seem to have survived in discrete refuges and have been rediscovered only recently. *X. apicalis* of which some vouchers from 1884 (Danube at Vienna) exist in the Museum of Natural History in Vienna was recently collected in the middle of the 16th district of Vienna (Graf, 2010). This long lost species is apparently recolonizing some large rivers in Central Europe (e.g. Braasch, 2003).

Few refugia of typical large river assemblages in Central Europe still exist, which can serve as reference sites. However, the records on theses reference assemblages are published in hidden taxonomic journals which are thus overseen by water authorities (e.g. Graf & Kovacs, 2002). The recent records of the species enable the identification and investigation of ecosystem processes, which support the occurrence of these species. One example for such an integer system is the River Lafnitz, a tributary of the Raba river, which has a length of 114 km and a catchment size of 1994 km², which is not seen as a 'large river' by the WFD. The Lafnitz still provides near-natural hydro-morphological conditions without impacts of navigation or big dams. Even though, the Lafnitz River cannot be considered as large river, it provides characteristic large river habitats.

Fish assemblages of the Danube

Fish assemblages of large rivers have been subject to human influence since millennia. In contrast to many other aquatic taxonomic groups biodiversity change was also initiated by direct human exploitation. For instance, on the Lower Danube fishery dates back to the Mesolithic and Neolithic period. In the Middle Danube fish remains from human settlements have been proven for the Late Copper Age about 4500 years before present (Bartosiewicz & Bonsall, 2004). Large predator species and sturgeons were main targets. Thus, exploitation and in recent centuries overexploitation was clearly an important driver of fish assemblage changes in large rivers beside the indirect effects of human induced habitat change and climate variability.

Materials

Fish assemblage changes of the Danube have been compiled for this report based on printed species inventories and scientific literature mainly of the 19th and 20th centuries as well as based on recent publications especially on fish catch statistics. For the Austrian Danube and Danube sturgeons also archival sources have been utilized. The systematic screening of commercial catch data as collected and reported by FAO since the 1950s (FAO Capture, Aquaculture and Global production databases 1950 – 2014) did not yield results relevant for species conservation or assessment as especially for the Danube countries and riverine freshwater fish species no differentiation between individual species was made.



Results & Discussion

Based on the different documents and sources finally used several trends of fish assemblage development can be described for the Danube:

Long-term, regionally progressing decline of specific species and final threat of these species in the entire river.

The main example for this development are the four diadromous Danube sturgeon species (Huso huso, Acipenser stellatus, A. gueldenstaedti, A. nudiventris). Beluga sturgeons (Huso huso) might have migrated in the past as far upstream as Regensburg (e.g. Schmall & Friedrich, 2014). It is however unclear, if the Bavarian Danube was ever an important spawning place. In the Austrian Danube, regular migration of sturgeons are proven from the 12th up until the 16th century. After the 16th century, the number of migrating individuals seems to have declined. An increasing fishing pressure in the Hungarian Danube might be a plausible explanation (see Balon, 1968). As records of the Viennese fish markets on sturgeon delivery from Hungary indicate, the abundance of sturgeons in the Hungarian Danube declined dramatically in the late 19th century (Haidvogl et al., 2013; 2014; Jungwirth et al., 2014). Although to the best of our knowledge this was so far not yet proven and/or published this decline of yields seems to have been the result of an increasing catch in the Lower Danube. At the end of the 19th century, average yearly catch of sturgeons in the Lower Danube amounted to about 700 tons. In the 20th century, human induced habitat alterations added to (over-)exploitation. In the 1960s, beluga, Russian and stellate sturgeon, as well as few ship sturgeons where caught regularly in Yugoslavia. Catches of about 16 tons per year reported in Busnita (1967), in 1961, e.g. 12 tons were caught, in 1962, 21 tons. From the 1970s onwards, the Iron Gate hydropower dams stopped spawning migrations beyond the Lower Danube. In the Lower Danube and in the delta populations dramatically declined until the end of the 20th century due to continued partly illegal fishing and habitat alterations aiming mainly at supporting navigation.

Fig. 23 depicts (average) catches of sturgeons in the Lower Danube and the Danube Delta since the late 19th century (Schiemer et al., 2003; internal data M. Staras, DDNI).



Fig. 23: Catches of sturgeons in the Lower Danube and in the Danube Delta according to Schiemer et al. (2003; right) and internal data of M. Staras (left)



Sturgeons and diadromous species in general are a valuable example to demonstrate the long-term effects and patterns of fish population decline in large rivers due to overfishing. A decline of sturgeon catch and average fish lengths from the 7th/9th to the 12/13th century in archaeological sites along tributaries of the Southern Baltic Sea demonstrates that the Danube was by no means a singular case (Hoffmann, 1996).

Extinct species with a narrow geographical niche

According to Kottelat & Freyhof (2007) 13 fish species have gone extinct in Europe since 1700. Notably, four are from the Danube catchment although definite species identification is in some cases difficult due to the lack of museums specimen available for morphological and genetic analysis. This concerns specifically *Alburnus danubicus* (Danube shemaya) which was recorded for the Lower Danube in Romania and Bulgaria and coastal lakes. Only two written descriptions from Antipa (1909) and Drensky (1943) exist as basis for species determination (see Kottelat & Freyhof, 2007). *Romanogobio antipai* (Danube delta gudgeon) occurred in the Lower Danube in Romania and Ukraine from the confluence with Arges downstream to the delta. *Salmo schiefermuelleri* (Mayforelle) was reported e.g. by Heckel (1851) and Heckel & Kner (1858) for subalpine lakes of the Austrian Danube catchment (e.g. Attersee, Traunsee, Fuschlsee) and occurred probably also in lakes of the Swiss Rhine drainage. *Gasterosteus crenobiontus* (Techirghiol stickleback) inhabited the freshwater springs of the hypersaline Romanian Lake Techirghiol. It is assumed that it has gone extinct due to the hybridization with *G. aculeatus* after the originally separated habitats of these two species were transformed and became connected due to irrigation.

Identifying the extinction of an individual species especially on regional scale strongly depends on sufficient sampling effort. For instance, in Austria the European mudminnow (*Umbra krameri*) was considered to be extinct after 1975. This species inhabits small, shallow and vegetated waterbodies of the Middle and Lower Danube and is native in the Upper Danube as far upstream as Vienna. Suitable habitats strongly decreased in the 20th century due to flood plain drainage and disconnection. Nevertheless, in Austria a remaining population was rediscovered by Wanzenböck in 1992 in a floodplain waterbody of the present national park downstream of Vienna (Wanzenböck, 1995; Wanzenböck & Spindler, 1995).

Decline of phytophilic spawners due to large scale disconnection of floodplains in the Middle and Lower Danube (and large tributaries)

As described by Staras (e.g. 1999) hydrology was an important factor of fish productivity in the Danube delta. Productivity increased with the number of days and height of flooding as reported by Anitpa (1912): In 1904/05 (1 April 1904-31 March 1905) no flood occurred and a total of 920 tons of fish were caught. In 1907/08 the water level reached 5.4 m and the inundation lasted for 128 days; a total of 6,448 tons fish were caught. Drainage measures halved the floodplains originally amounting to 466,000 ha in the late 1950s and continued in the 1960s. This also altered



the hydrological regime. As a result fish productivity declined. Total fish catch declined from 10-20.000 tons per year before and during the 1960s to 5,000 - 6,000 tons after 1984. This development affected species differently. Especially for pike-perch (*Sander lucioperca*) and pike (*Esox lucious*) as phytophilic spawners catches dropped in the late 1960s from 2,500 tons to about 500 tons in the 1970s and less after (for pike) and from up to 300 tons in the 1960s to 100 tons and less afterwards (for pike-perch; see Fig. 22). Catfish (*Silurus glanis*), a nest guarder, was effected similarly in the delta.



Fig. 24: Catches of pike (left), pike-perch (middle) and catfish (right) in the Lower Danube and Danube delta from 1960 – 1993 (data from Staras, DDNI)

A similar link between hydrological conditions and fish catch has been reported for the Tisza at Szeged already in the 19th century (1834 to 1900). Also here fish ecological conditions were fundamentally altered due to channelization of the river between 1846 and the 1880s aiming among others to colonize the extensive floodplains for grain production as new railway lines offered the opportunity to export them into other parts of the Austro-Hungarian monarchy. As a consequence, 1.6 Mio ha floodplains were lost and only 0.2 Mio ha remained. Fish catch declined drastically although unfortunately no statistics have been published but only a verbal assessment of the situation (Repassy, 1906).

(Intended) introduction of non-native fish

Non-native fish species had been introduced to the Danube since more than 100 years for commercial fishery purposes (eel) or recreational fishery (rainbow trout or brown bullhead), for water management (grass carp, silver carp) or they occur due to invasive behavior (Neogobiidae, see below). Altogether, 14 species were purposefully introduced along the entire Danube during the 20th century (Schiemer et al., 2003). According to Schiemer et al. (2003), two of them have to be considered as naturalized: false harlequin (*Pseudorasbora parva*) and pumpkinseed (*Lepomis gibbosus*). If Prussian carp (*Carassius gibelio*) is perceived as non-native a third species is acclimatized. However, the status of this species in the Danube catchment seems still unclear (see also Kottelat & Feyhof, 2007). It appeared in the Lower Danube in the first quarter of the 20th century but was rare until 1970. Soon after, it invaded the middle and the upper sections of the river. In the Romanian Danube delta the Prussian carp (*Ctenopharyngodon idella*) was introduced in fish farms in 1962 to increase productivity and to control aquatic vegetation in ponds. It has reproductive populations in the lower Danube, as is also the case for silver carp (*Hypothalmichthys molitrix*) and bighead carp (*Aristichthys nobilis*) (Schiemer et al. 2003).



Recent occurrence of invasive fish species

Invasive fish of the Danube relates in particular to the human induced, unintended dispersal of species which are native in the Lower Danube and spread into the Middle and Upper Danube and in recent years also to tributaries such as the Austrian Morava and Thaya (Lusk et al., 2010). Dispersal happened mainly with the release of ballast water from ships as the first appearances of these species were observed in and close to the large freight harbors (Wiesner, 2005). For fish of the (former) Neogobiidae family, no records upstream of the Iron Gate were made before 1970. In 1994 they were observed for the first time in the upper Danube (Austria; Wiesner, 2005). At present, *Neogobius melanostomus* (round goby) and *Ponticola kessleri* (former *N. kessleri*) became dominating in the rip-rap all along the Danube between Kelheim up to the Iron Gate. In the 2nd Joint Danube Survey (JDS), round goby was the third most common species after bleak (*Alburnus alburnus*) and Prussian carp and *P. kessleri* was the fifth most common after roach (*Rutilus rutilus*). In JDS 3 apart from bleak, round goby clearly dominated the fish community with a relative proportion of 46 % and 26 %, respectively, whereas the round goby was in particular frequent in the Danube upstream the Iron Gate where it is non-native (Bammer et al., 2015).

Fishery catch statistics and patterns of fish assemblage and populations decline

It is often difficult to make direct use of fishery catch data as long-time series are missing and/or fishing effort is insufficiently reported. However, data from the Danube delta investigated by fish biologists from the DDNI (e.g. Navodaru et al., 2001) and data analysed by Serbian fish biologists (esp. Jaric et al., 2016) demonstrate the interest of catch data to reveal patterns of human pressure on freshwater fish assemblages as this has been proven already for marine fish and fishery (e.g. Pauly and Palomares, 2005; Essington et al., 2006). In the Danube Delta, the catch of catfish, pike and pike-perch declined in the beginning of the 1970s. After, yields especially of Prussian carp and common bream (Abramis brama) increased putting more intense fishing pressure on these species (Fig. 25). Although a decline of individual species in commercial fishery has to be interpreted with caution as cultural aspects can be involved, too, this example seems to reflect a real species decline rather than a change of preferences for other species. For the Serbian Danube commercial catch are available since the late 1960 reporting among others yields for catfish, pikeperch, carp (Cyprinus carpio), Prussian carp and common bream. As Jaric et al. (2016) point out fish catch statistics were particularly affected by misreporting in the period of 1990 -2005 and thus omitted in their analysis. In the two periods finally kept (1969-1989; 2006-2010) total catch in tons did not decline as much as in the Danube Delta and in fact even increased in recent years. But more detailed analyses of e.g. average fish lengths and weights as well as mean trophic level of caught fish showed that fishery has unsustainable effects as within the studied period fish tended to become smaller, to reproduce younger and to have a shorter lifespan.



Deliverable 5.C Report on legacy and tipping points in large rivers



Fig. 25: Increase of catches of Prussian carp and common bream after the decrease of pike and pikeperch harvest in the Lower Danube and Danube delta



Synthesis

The ecological condition of large rivers depend on drivers and pressure and resulting stressors that can occur at multiple spatial scales (Frissell et al., 1986; Ward et al., 2002; Wiens, 2002). However, the ecological condition of a large river is not only dependent on the local condition of the river system but also on stressors operating at larger spatial scales as the catchments of large rivers cover large areas (Allan et al., 1997; Roth et al., 1996). Accordingly, the stressor may evolve over longer temporal scales. Accordingly, investigating the changes which occurred over long periods may give valuable insights and deepen the understanding of assemblage shifts which in turn represent the quintessential basis for bio-assessment (Birk et al., 2012).

Large rivers have been altered since centuries. Hering et al. (2015) summarise the multiple interaction between various stressors of aquatic ecosystems worldwide. The rates of habitat modification in large rivers have been so high that virtually all natural habitats and protected areas are destined to become ecological 'islands' in surrounding 'oceans' of altered habitats. This process of fragmentation and isolation in landscapes under human influence – main concepts in island biogeographic theory – is predicted to lead directly and indirectly to accelerated species extinctions at both the local and the global scale, thus reducing the world's biodiversity at all levels (McArthur & Wilson, 1967; Lawton & May, 1995).

Potamal communities at the edge of their ecological capability might collapse when temperature increase due to climate change adds to the deadly anthropogenic cocktail (Travis, 2003). But with few exceptions there is no evidence of an actual decrease in species richness of rather flexible riverine and wetland assemblages in lowlands of Central Europe, simply because most of these communities have been already dramatically shaped by various anthropogenic pressures in the decades before; those surviving organisms are tolerant cosmopolitans which cover a large area of ecoregions.

There are few signals of a re-colonisation regarding some riverine species which indicate improvements in the overall ecological integrity, mainly water quality. Awareness of the vulnerability and sensitivity of the large river ecosystems has risen and various restoration plans are put in praxis like along the Danube. River systems incorporate processes within the entire catchment and local efforts - despite their undoubted merits - can only marginally soften large scale impairments such as disruption of migration pathways or continuous channelization.

Biological reference communities are lost for most large rivers since long times and cannot be described empirically (e.g. Ehlert et al., 2002). Large rivers in Europe have lost a high share of their indigenous fauna, especially sensitive insects like Ephemeroptera, Plecoptera and Trichoptera (EPT-taxa), which is underlined for Plecoptera by the presented results. The findings of this study improve the knowledge on the historical change of riverine macroinvertebrates and give important evidence on the faunal shifts which occurred in large European rivers. In contrast to fish, which have been of economic interest and at least fisheries statistics give hints on historical assemblages, there are few studies dealing with historical occurrences of macroinvertebrate species in rivers.



We found a considerable shrink in the area of selected indicator species, namely of the aquatic insect order Plecoptera, especially for the co-occurrence of the selected species; the main refugia are few aquatic systems in Central France (Loire, Allier), Austria/Hungary (Raba, Lafnitz) and Hungary/Romania (Tisza) as well as in remote areas of Eastern Europe (Lithuania, Belarus), where a combination of species still can be found after 1990. Theses river systems are still inhabited by a high and typical fauna and seem to be least disturbed systems. Losses are identified in many rivers, especially in Scandinavia and Spain as well as other southern European regions.

In fact, many typical and nowadays extinct or endangered species of large rivers showed mass emergences and short but synchronic flight periods (e.g. *X. apicalis, I. obscura*). This phenomenon seems to be essential for mating and reproduction success; as minimum populationsize is not known, slight reductions of swarming stages may lead to severe bottlenecks leading to abrupt species losses within the whole catchment.

Large river systems with reduced occurrence of these species are multiple stressed by channelization, damming, navigation and neozoa. The relationships between these stressors and the major ecosystem responses are synthesised in Table 4. The combination of these stressors may lead in most cases to additive or synergistic effects. However, this interaction is not quantifiable from the available historical data for several reasons.

Table 4: Summary of relationships between pressures, impacts and their biotic and abiotic responses

Pressure	Impact	Abiotic response	Biotic response
Damming	Longitudinal Connectivity	Discontinuity	Migration
Damming	Hydrological/ sedimentological regime	Disturbance regime change / homogenisation	?
Damming	Sediment retention	River bed incision	
Damming	Flow reduction	Sediment retention / Siltation	Loss rheobiont species
Channelization	Lateral connectivity	Loss of wetlands	Species loss
Channelization	Increase of hydraulic stress	Loss of instream lentic habitats Loss of retention	Loss of stagnophiluous species – less shredders
Channelization	Rip rap	Homogenisation – loss of small-sized substrate Change land-water surface interface	Loss of fine sediment habitat
Navigation	Vessel-induced waves	Pulse release – lateral hydraulic stress	Physical disturbance of surface dwelling taxa – merolimnic species
Navigation	Morphological degradation navigation channel	Gravel extraction	Loss of habitat – extreme conditions
Navigation	Neozoa	Habitat engineering	Loss of habitat – less grazers Competition

In most cases the historical biological data is simply too scattered in time and space to gain a homogenous information and to compare different rivers or river sections over time. The initial



intent behind the work of this study has been to describe the development of stressors as well as faunal changes over time for several time slices within the 20th century. However, the collection of historical information is constrained by a two-sided challenge: detailed but small-scaled information vs. coarse but large-scaled information. To collate an appropriate database would take tremendous efforts, which are absolutely not feasible within this deliverable.

Looking in more detail on the species records, inconsistencies are obvious: The first period in the analyses till 1990 covers a much longer time frame (more than 100 years) compared to the period after 1990 (25 years). In contrast, the intensity of ecological sampling extremely increased in the 21st century induced by the WFD and its monitoring obligations. Furthermore, several ecologists even started campaigns to sample some of these species again without success. This information is written down in reports and articles but it is hardly possible to integrate it in a spatial explicit manner as in most cases this absence is described for large areas like whole countries or river systems. Large datasets with repeated samplings as compiled by the Joint Danube Surveys are extremely useful and necessary in monitoring of the ecosystem functioning and potential changes in ecosystem services.

Moreover, the stressor timelines underlined the parallel evolvement of the major stressors damming and navigation accompanied by channelization and neozoa in the second half of the 20th century, which impedes the untangling of detailed cause-effect relationships or the interaction types of stressors, respectively. Moreover, the quantifiable stressors do not represent the direct impact on the ecosystem process. They represent proxies, which can have similar effects on the ecosystem processes; in most cases an effect on the disturbance regime and in turn on the naturally dynamic habitat processes. Looking on river systems, which still feature some of the Plecoptereaen species of interest, it gets obvious that those systems show neither damming nor navigation impacts. Accordingly, dynamic processes in hydrology and sedimentology seem key to support large river typical Plecoptera (see Fig. 26).



Fig. 26: The Loire (left and middle) and Allier river (right), source: A. Ruffino

In relation to riverine ecosystem dynamics and especially for large rivers with adjacent floodplains habitat age gives the possibility to highlight the degree of dynamics in the system. In a pristine, least disturbed river-floodplain system, habitats along a hydrological gradient – from the lotic main channel to lentic floodplain lakes - can be found (Amoros et al., 1987). Channelization, damming and navigation reduce either the area where the dynamic processes can take place such as reducing lateral connectivity or they directly suppress them, e.g. by affecting the hydrological



regime. The changes of such habitat types are depicted in Fig. 23 for the Danube at Vienna (a channelized, dammed and navigated river section) from the beginning of the 19th century up to 2011 (Graf et al., 2013). If the hydrological dynamics in the floodplain are completely eliminated the remaining lentic habitats are exposed to siltation and will diminish over time.



Fig. 27: Habitat types along the Danube River at Vienna in 1817 (left) and in 2011 (right); the classification of habitats distinguishes hydrological gradients according to Amoros et al. (1987) from Graf et al. (2013)

In the context of the so-called 'McDonaldization' of the biosphere (Lövei, 1997) the dispersal of many species is inhibited, while others - mostly more flexible species in ecological terms - become common and overtake the niches of indigenous species. The role of navigation in the McDonaldization-process is hardly investigated comprehensively in all its aspects, therefore poorly understood and still remains underestimated. Replacement of vulnerable taxa by rapidly spreading taxa that thrive in human-altered environments will ultimately produce a spatially more homogenised biosphere with much lower diversity. Regarding aquatic ecosystems and in particular large rivers, similar processes have already been observed by Fittkau & Reiss (1983), Zwick (1984; 1992) and Fochetti & Tierno de Figueroa (2006). Caspers reported already in 1980 that the benthic macroinvertebrates coenosis in the Rhine River, which can be seen as the 'pioneer' system of human-induced alterations, can be characterised as uniform and species-poor assemblage at Bonn. As pointed out earlier, the benthic assemblages are nowadays clearly dominated by non-indigenous, invasive or cosmopolitan elements which probably have strong negative effects and misbalance the ecological functioning of the whole system (Fig. 24).

Deliverable 5.C Report on legacy and tipping points in large rivers





Fig. 28: Concept on the development of the large river fauna in Central Europe from 1800 to 2000; photos: left, indigenous species of the Danube: B. trifasciata, X. apicalis, T. araneoides (pinned specimen, Museum Budapest, Photo: D. Murányi); right, invasive species: Corbicula fluminea, Dikerogammarus villosus, Chelicocoropium curvispinum (Graf & Pletterbauer, unpublished)

The seriousness of the neozoa problem may be illustrated exemplarily by the recently documented structure of benthic assemblages of the Danube River during the Joint Danube Survey 2 (ICPDR 2008) expeditions which sampled along the whole river: Among the ten most frequent macroinvertebrate species sampled, nine are assigned as neozoa (Graf et al., 2008), above all occurring in very high densities and frequency (Fig. 13). In general these findings were confirmed by the JDS3 expedition (Borza et al., 2015). Detailed analyses of JDS data reveal clear habitat preferences for hard substrates for invasive species and soft bottom preferences for non-invasives, respectively (Borza et al., subm.). Land-water interface of large rivers is mostly interrupted by rip-rap structures. In combination with vessel-induced waves, which are especially harmful for merolimnic taxa, they seem to be support the success of invasiveness of Ponto-Caspian species.

Deliverable 5.C Report on legacy and tipping points in large rivers





Fig. 29: Cumulative abundance (boxplots, in individual/m2) of invasive (A) and non-invasive (B) Ponto-Caspian peracarid species on different substrate types downstream of rkm 700 in the Danube during Joint Danube Survey 3 (Borza et al., subm.)



References

- Allan JD (1995) Stream ecology. Structure and function of running waters. Chapman & Hall, London, 388 pp.
- Allan JD, Erickson DL, Fay J, (1997) The influence of catchment land use on stream integrity across multiple spatial scales. Freshw. Biol. 37, 149–161. doi:10.1046/j.1365-2427.1997.d01-546.x
- Amoros C, Roux AL (1988) Interaction between water bodies within the floodplains of large rivers: function and development of connectivity. Munst. Geogr. Arb. 29, 125–130.
- Amoros C, Roux AL, Reygrobellet JL, Bravard JP, Pautou G (1987). A Method for Applied Ecological Studies of Fluvial Hydrosystems. Regul. Rivers 1, 17–36. doi:10.1002/rrr.3450010104
- Antipa G (1909) Fauna ichtiologică a româniei. Publicatiunile fondului Vasile Adamachi; no. XVI. Academia Romana Bucuresti
- Antipa G (1912) Die Biologie des Inundationsgebiets der Unteren Donau und des Donaudeltas. In: Stummer-Traunfels R (ed) Verhandlungen des VIII Internationalen Zoologen Kongresses zu Graz vom 15.-20. August 1910. G. Fischer, Jena: 163-208
- Ausserer C (1869) Neurotteri tirolesi colla diagnosi di tutti i generi Europei. Parte I. Pseudoneurotteri. Annuario della Societa di Naturalisti in Modena. Modena 4:71–156
- Bacela K, Grabowski M, Konopacka A (2008) Dikerogammarus villosus (Sowinsky, 1894) (Crustacea, Amphipoda) enters Vistula the biggest river in the Baltic basin. Aquat Invasions 3(1):95–98
- Balon E (1968) Einfluß des Fischfangs auf die Fischgemeinschaften der Donau. Archiv für Hydrobiologie/Supplement, 34(3): 228-249
- Bammer V, György A, Pehlivanov L, Schabuss M, Szaloky S., Zornig H (2015) Fish. In: Liska I, Wagner F, Sengl M, Deutsch K, Slobodnik J (eds) Joint Danube Survey 3. ICPDR Vienna: 126-139
- Banning M (1998) Auswirkung des Aufstaus größerer Flüsse auf das Makrozoobenthos dargestellt am Beispiel der Donau. Essener Ökologische Schriften, Bd. 9:1–185. Westarp Wissenschaften
- Bartosiewicz L, Bonsall C (2004) Prehistoric fishing along the Danube. Antaeus 27: 253-272
- Bij de Vaate A, Jazdezewski K, Ketelaars HAM, Gollach S, Van der Velde G (2002) Geographical patterns in range extension of Ponto-Caspian macroinvertebrate species in Europe. Can J Fish Aqaut Sci 59:1159–1174
- Birk S, Bonne W, Borja A, Brucet S, Courrat A, Pokane S, Solimini A, van de Bund W, Zampoukas N, Hering D (2012) Three hundred ways to assess Europe's surface waters: An almost complete overview of biological methods to implement the Water Framework Directive. Ecol. Indic. 18, 31–41. doi:10.1016/j.ecolind.2011.10.009
- Böhme D (2004) Rote Liste der Eintags- und Steinfliegen (Ephemeroptera, Plecoptera) des Landes Sachsen-Anhalt. Berichte des Landesamtes für Umweltschutz Sachsen-Anhalt 39: 198-203.
- Bojkova J (2009) Revision of the stonefly collections (Plecoptera) by E. Křelinová and J. Raušer from the Czech Republic. Aquatic Insects 31(suppl. 1): 245–251.
- Bojkova J, Chvojka P & Komzak P (2012a) Stoneflies (Plecoptera) of the Bílé Karpaty Protected Landscape Area and Biosphere Reserve (Czech Republic). Acta Musei Moraviae, Scientiae Biologicae 96(2) (2011): 37–70.
- Bojkova J, Komprdova K, Soldan T & Zahradkova S (2012b) Species loss of stoneflies (Plecoptera) in the Czech Republic during the 20th century. Freshwater Biology 57: 2550–2567.
- Bojkova J, Radkova V, Soldan T. & Zahradkova S (2013): Trends in species diversity of lotic stoneflies (Plecoptera) in the Czech Republic over five decades. Insect Conservation and Diversity, in press, DOI: 10.1111/icad.12050.
- Bojkova J, Soldan T, Špacek J & Straka M. (2011) Distribution of stoneflies of the family Taeniopterygidae (Plecoptera) in the Czech Republic: earlier data, new records and recent distributional changes. Časopis Slezského Muzea Opava, Série A – Vědy Přírodní 60: 239–258.



- Borza P, Huber T, Leitner P, & Graf W (submitted): Need for speed: co-existence patterns of invasive *Dikerogammarus* spp. (Crustacea: Amphipoda) explained by niche differentiation along the current velocity gradient.
- Borza P, Csanyi B, Huber T, Leitner P, Paunovic M, Remund N, Szekeres J & Graf W (2015) Longitudinal distributional patterns of Peracarida (Crustacea, Malacostraca) in the River Danube. Fund Appl Limnol. 2015; 187(2): 113-126
- Boumans L (2011) Databasing the stoneflies (Plecoptera) at the Natural History Museum in Osloreveals new Norwegian province records. Norwegian Journal of Entomology 58, 170–179.
- Braasch D & Berger T (2003) Rote Liste und Artenliste der Steinfliegen (Plecoptera) des LandesBrandenburg) Naturschutz und Landschaftspflege in Brandenburg Beilage zu Heft 4.
- Braasch D (2003) Rückkehr der Steinfliegen (Plecoptera) in Oder und Lausitzer Neiße. Lauterbornia 46:93–101
- Brauer F & Löw F (1857) Neuroptera austriacus. Die im Erzogthum Oesterreich bis jetzt aufgefundenen Neuropteren nach der analytischen Methode zusammengestellt, nebst einer kurzen Charakteristik aller europäischen Neuropteren Gattungen. Verlag Carl Gerold's Sohn, Perlidae, pp 27–31
- Bravard JP, Amoros C & Pautou G (1986) Impact of civil engineering works on the succession of communities in a fluvial system. Oikos 47: 92-111.
- Busnita T (1967) Die wirtschaftliche Bedeutung der Donau. Die Fischerei und Fischwirtschaft. In: Liepolt R (ed) Limnologie der Donau. Schweizerbart´sche Verlagsbuchhandlung, Stuttgart, Lieferung 4: 26-41
- Calderini P (1869) Apparizione di un numero straordinario di Nemure Nebulosa (Insetti Neurotteri) nei dintori di Varallo. Tip Colleoni (Varallo) 1–5 129.
- Chovet M. & Lecureuil J (2009). The macrobenthos in the middle part of the Loire catchment (France) and its evolution since 1977. Ephemera 10: 103-22.
- Craig R (2015) A review of the stoneflies (Plecoptera) of Great Britain Species Status No.20. Natural England Commissioned Report NECR174.
- Cummins KW, Sedell JR, Swanson FJ, Minshall GW, Fisher, SG, Cushing CE, Petersen RC & Vannote RL (1983) Organic matter budgets for stream ecosystems: problems in their evaluation. In: Barnes, JR & Minshall GW (eds.): Stream ecology: application and testing of general ecological theory. Plenum Press, New York-London, pp. 299–353.
- Demek J, Drescher A, Hohensinner S & Schwaighofer B (2008) The geology and geomorphology of floodplains. In: Klimo E, Hager H, Matic S, Anic I, Kulhavy J (eds) Floodplain forests of the temperate zone of Europe. Lesnicka prace, Kostelec nad Cernymilesy, pp 11–38
- Ehlert T, Hering D, Koenzen U, Pottgiesser T, Schuhmacher H, Friedrich G (2002) Typology and type specific reference conditions for medium-sized and large rivers in North Rhine-Westphalia: Methodical and biological aspects. Int. Rev. Hydrobiol. 87, 151–163. doi:10.1002/1522-2632(200205)87:2/3<151::AID-IROH151>3.0.CO;2-A
- Essington TE, Beaudreau AH, Wiedenmann J (2006) Fishing through marine food webs. Proceedings of the National Academy of Sciences of the United States of America 103(9): 3171-3175
- Fialkowksi W & Sowa R (2002) Plecoptera Stoneflies. In: Glowacinski Z (ed.): Red List of threatened animals in Poland. Polska Akademia Nauk, Instytut Ochrony Przyrody, Kraków, pp. 122–124. (in Polish with English summary)
- Fittkau EJ & Reiss F (1983) Versuch einer Rekonstruktion der Fauna europäischer Ströme und ihrer Auen. Arch Hydrobiol 97(1):1–6
- Fochetti R & Tierno de Figueroa JM (2008) Global diversity of stoneflies (Plecoptera; Insecta) in freshwater. Hydrobiologia 595:365–377
- Frissell C, Liss W, Warren C, Hurley M (1986). A hierarchical framework for stream habitat classification: Viewing streams in a watershed context. Environ. Manage. 10, 199–214. doi:10.1007/BF01867358



- Gabel F, Garcia XF, Brauns M, Sukhodolov A, Leszinski M & Pusch MT (2008) Resistance to shipinduced waves of benthic invertebrates in various littoral habitats. Freshw Biol 53:1567–1578
- Gabel F, Pusch MT, Breyer P, ter Burmes V, Walz N & Garcia XF (2011) Differential effect of wave stress on the physiology and behaviour of native versus non-native benthic invertebrates. Biol Invasions 13:1843–1853
- Gabel F, Stoll S, Fischer P, Pusch MT & Garcia XF (2011) Waves affect predator–prey interactions between fish and benthic invertebrates. Oecologia 165:101–109
- García de Jálo D, Alonso C, Gonzalez del Tango M, Martinez V, Gurnell A, Lozrenz S, Wolter C, Rinaldi M, Belletti B, Mosselman E, Hendriks D & Geerling G (2013) Review on pressure effects on hydromorphological variables and ecologically relevant processes. Deliverable D1.2 "Effects of pressures on hydromorphology" REFORM.
- Geerling GW (2008) Changing Rivers: Analysing fluvial landscape dynamics using remote sensing. PhD thesis, Centre for sustainable management of resources, Radboud University of Nijmegen, The Netherlands.
- Gergel SE, Turner MG, Miller JR, Melack JM, Stanley EH (2002) Landscape indicators of human impacts to riverine systems. Aquat. Sci. Across Boundaries 64, 118–128
- Graf W (2010) Eine aktualisierte Checkliste der Steinfliegen (Insecta: Plecoptera) Österreichs. Lauterbornia 71:175–183
- Graf W, Bloch A & Moog O (2005) Vergleichstudie March im Abschnitt Marchegg (Flusskm 15,00-25,00) Teilbereich Makrozoobenthos. Endbericht im Auftrag des Umweltbundesamtes, Wien
- Graf W, Chovanec A, Hohensinner S, Leitner P, Schmidt-Kloiber A, Stubauer I, Waringer J & Ofenböck G (2013) Das Makrozoobenthos als Indikatorgruppe zur Bewertung großer Flüsse unter Einbeziehung auenökologischer Aspekte Österreichische Wasser- und Abfallwirtschaft, 65, 386-399; ISSN 0945-358X
- Graf W, Murphy J, Dahl J, Zamora-Munoz C & Lopez-Rodriguez MJ (2008) Distribution and Ecological Preferences of European Freshwater Organisms - Volume 1. Trichoptera. In: A. Schmidt-Kloiber and D. Hering (eds.): Distribution and Ecological Preferences of European Freshwater Organisms. Pensoft Publishers (Sofia-Moscow), 388pp.
- Graf, W., Neu, P. J., and Schmidt-Kloiber, A. (eds.), 2016, Distribution Atlas of European Plecoptera (DAEP)
- Graf W, Kovacs T (2002) The aquatic invertebrates of the Lafnitz-Raba river system in Austria and Hungary- a natural heritage of the Central European potamocoen. Internat Assoc Danube Res Tulcea 34:295–303
- Greulich, S. (2014): Mise en place d'un réseau d'Observation de la Biodiversité de la Loire et de ses Affluents (OBLA). Rapport scientifique UMR Université de Tours CNRS 7324 CITERES.
- Grossinger L (1794) Universa historia physica regni Hungariae, secundum tria regna naturae digesta. III. Ichtyologia sive Historia Piscium et Amphibiorum. Posonii et Comaromi XIV:408
- Habersack H, Jäger E, Hauer C (2013) The status of the Danube River sediment regime and morphology as a basis for future basin management. Int J River Basin Manage 153–166
- Haidvogl G, Galik A, Pont D (2013) Vers une histoire de la biodiversité piscicole du Danube autrichien : apports croisés des sources archéologiques et écrites. Revue Du Nord: 131-142
- Haidvogl G, Lajus D, Pont D, Schmid M, Jungwirth M, Lajus J (2014) Typology of historical sources and the reconstruction of long-term historical changes of riverine fish: a case study of the Austrian Danube and northern Russian rivers. Ecology of Freshwater Fish 23: 498-515
- Heckel JJ (1851) Bericht einer auf Kosten der kaiserlichen Akademie der Wissenschaften durch Oberösterreich nach Salzburg, München, Innsbruck, Botzen, Verona, Padua, Venedig und Triest unternommenen Reise. (Sitzung vom 24.Juli 1851). Sitzungsberichte der mathematischnaturwissenschaftlichen Classe der kaiserlichen Akademie der Wissenschaften. Wien, Kaiserlichkönigliche Hof- und Staatsdruckerei. 7: 281-332.



- Heckel JJ, Kner R (1858) Die Süsswasserfische der östreichischen Monarchie mit Rücksicht auf die angränzenden Länder. W. Engelmann, Leipzig
- Hering D, Carvalho L, Argillier C, Beklioglu M, Borja A, Cardoso AC, Duel H, Ferreira T, Globevnik L, Hanganu J, Hellsten S, Jeppesen E, Kodes V, Solheim AL, Noges T, Ormerod S, Panagopoulos Y, Schmutz S, Venohr M, Birk S (2015) Managing aquatic ecosystems and water resources under multiple stress - an introduction to the MARS project. Sci Total Environ 15(503–504):10–21
- Herzig A, Bretschko G, Gaviria EA, Stipanits B, Zoufal W (1987) Quantitaive Zoobenthosuntersuchungen im Stauraum Altenwörth. 26. Arbeitstagung der IAD, Passua/Deutschland, pp 127–132
- Hirzinger V, Bartl E, Weissenbacher A, Zornig H, Schiemer F (2002) Habitatveränderungen durch den schiffahrtsbedingten Wellenschlag und deren potentielle Auswirkung auf die Jungfischfauna in der Donau. Österreichs Fischerei 55:238–243
- Hoffmann RC (1996) Economic Development and Aquatic Ecosystems in Medieval Europe. The American Historical Review 101(3): 631 669
- Hohensinner S (2008) Rekonstruktion ursprünglicher Lebensraumverhältnisse der Fluss-Auen-Biozönose der Donau im Machland auf Basis der morphologischen Entwicklung von 1715 1991. Dissertation an der Universität für Bodenkultur Wien, 307 S. u. 27 Kartenbeilagen
- Hohensinner S, Habersack H, Jungwirth M, Zauner G (2004) Reconstruction of the characteristics of a natural alluvial river-floodplain system and hydromorphological changes following human modifications: the Danube River (1812–1991). River Res Appl 20(1):25–41
- Hohensinner S, Jungwirth M, Haidvogl G, Muhar S, Preis S, Schmutz S (2005) Historical analysis of habitat turnover and age distributions as a reference for restoration of Austrian Danube floodplains. River Basin Management III, WIT Transactions on Ecology and the Environment, vol 83. WIT Press, Ashurst, pp 489–502. ISBN 1-84564-023-3. ISSN 1746-448X
- Hohensinner S, Jungwirth M, Muhar S, Schmutz S (2011) Spatio-temporal habitat dynamics in a changing Danube River landscape 1812-2006. River Res. Appl. 27, 939–955. doi:10.1002/rra
- Hohensinner S, Herrnegger M, Blaschke AP, Habereder C, Haidvogl G, Hein T, Jungwirth M, Weiß M (2008) Type-specific reference conditions of fluvial landscapes: a search in the past by 3D-reconstruction. Catena 75:200–215, ISSN 0341–8162
- Hughes FMR, Colston A & Mountford JO (2005) Restoring riparian ecosystems: the challenge of accommodating variability and designing restoration trajectories. Ecology and Society 10(1): 12.
- ICPDR (2008) Joint Danube Survey 2 final scientific report, Liska I., Wagner F., Slobodnik J. (eds.), 242pp
- Jarić I, Smederevac-Lalić M, Jovičić K, Jaćimović M, Cvijanović G, Lenhardt M, Kalauzi A (2016) Indicators of unsustainable fishery in the Middle Danube. Ecology of Freshwater Fish 25: 86-98
- Jepsen N, Wiesner C, Schotzko N (2008) Fish. In: Liška I, Wagner F, Slobodník J (eds), Joint Danube Survey. Final scientific report. Vienna: 72-81.
- Johnson WC (1997) Equilibrium response of riparian vegetation to flow regulation in the Platte river, Nebraska. Regulated Rivers: Research & Management 13: 403-415.
- Jungwirth M, Haidvogel G, Hohensinner S, Waidbacher H & Zauner G (2014) Österreichs Donau. Landschaft – Fisch – Geschichte. Institut für Hydrobiologie und Gewässermanagement, BOKU Wien, p 420
- Junk WJ, Bayley PB, Sparks RE (1989) The flood pulse concept in river-floodplain systems. Can. Spec. Publ. Fish. Aquat. Sci. 110–127.
- Kjærstad G, Andersen T, Olsvik HA & Brittain, JE (2010) Døgnluer, øyenstikkere, steinluer og vårluer. Pp. 227–234 in Kålås, J.A., Viken, Å.,Henriksen, S. & Skjelseth, S. (Eds.) Norsk rødliste for arter 2010. Artsdatabanken, Trondheim.
- Klink AG (1989) 11. The Lower Rhine: Palaeoecological analysis. In: Petts, G., (ed.). Historical changes in large alluvial rivers: Western Europe. Wiley and Sons Ltd. 183-201.



- Kohl W (2010) Von der Unsauberkeit in Wien und deren Beseitigung im 16. und 17. Jahrhundert. Wiener Geschichtsblätter 147–191
- Korte T, Sommerhäuser M (2012) Auswirkungen des Klimawandels auf die Bewertung "Großer sand- und lehmgeprägter Flüsse des Tieflands". NW Korrespondenz Wasserwirtschaft 6:309–315
- Kottelat M, Freyhof J (2007) Handbook of European Freshwater Fishes. Eigenverlag, Cornol
- Kovacs T, Ambrus A, Juhasz P (2002) Ephemeroptera, Odonata, and Plecoptera larvae from River Tisza in the year of cyanide pollution (2000). Folia Historica Naturalia Musei Matraensis 26:169–178
- Kovács T, Graf, W & Ambrus A (2004) Besdolus ventralis (Pictet, 1841) and Isogenus nubecula Newman, 1833 from the Austrian reaches of the Lafnitz river (Plecoptera: Perlodidae). Folia ent. hung. 65: 33-36.
- Kucera-Hirzinger V, Schludermann E, Zornig H, Schabuss M & Schiemer F (2009) Potential effects of navigation-induced wave wash on the early life history stages of riverine fish. Aquat Sci 1(1):94–102
- Kühtreiber J (1934) Plekopterenfauna Nordtirols. Sonderabdruck aus den Berichten des Naturwissenschaftlich-Medizinischen Vereines in Innsbruck, XLIII./XLIV. Jahrgang (1931/32 bis 1933/34):1–214
- Lake PS (2000) Disturbance, patchiness, and diversity in streams. J. North Am. Benthol. Soc. 19, 573–592.
- Lawton JH, May RM (eds) (1995) Extinction rates. Oxford University Press, Oxford
- Liebmann H, Reichenback-Klinke H (1967) VI. Eingriffe des Menschen und deren biologische Auswirkung. Limnologie der Donau, Liefg 4: 1–25
- Liedermann M, Tritthart M, Gmeiner P, Hinterleitner M, Schludermann E, Keckeis H., Habersack H (2014) Typification of vessel-induced waves and their interaction with different bank types, including management implications for river restoration projects. Hydrobiologia 729, 17–31. doi:10.1007/s10750-014-1829-1
- Lövei, GL (1997) Global change through invasion. Nature 388, 627–628.
- Lusk S, Luskova V, Hanel L (2010) Alien fish species in the Czech Republic and their impact on the native fish fauna. Folia Zoologica 59(1): 57-72
- Marston RA, Girel J, Pautou G, Piegay H, Bravard J-P, Arneson C (1995) Channel metamorphosis floodplain disturbance, and vegetation development: Ain river, France. Geomorphology 13: 121-131.
- Mauch E (1981) Der Einfluss des Aufstaus und des Ausbaus der deutschen Mosel auf das biologische Bild und den Gütezustand. DVWK-Schriften 4:39–137
- McArthur RH, Wilson EO (1967) The theory of island biogeography. Princeton University Press, Princeton
- Middelkoop, H, MM Schoor, HP Wolfert, GJ Maas & E Stouthamer (2005) Targets for ecological rehabilitation of the lower Rhine and Meuse based on a historic-geomorphologic reference. Archiv fur Hydrobiologie Supplement 115/1-4; Large Rivers Volume 15, No. 1-4: page 63-88.
- Moog O (1986) Überlegungen zur Gütebeurteilung von Flussstauen. Fallstudie am Beispiel des Traunstaues Pucking. In: Moog O (1990) Auswirkungen anthropogener Eingriffe auf aquatische Ökosysteme. Habilitationsschrift, Univ. f. Bodenkultur, Wien, pp 1–110
- Moog O, Jungwirth M (1992) Wasserkraft-Nutzung und Restwassermengen mit besonderer Berücksichtigung ökologischer Gesichtspunkte. Tagungsberichte 17. Flußbautagung 28.9.- 2.10.1992, Bregenz
- Moog O, Wieser C (2010) Meeresgrundel, Körbchenmuschel, Schwebgarnele und Co. Gebietsfrenmde Fische und Bodentiere in Österreichs Flüssen und Seen. In: Rabitsch W, Essl F (eds) Aliens –Pflanzen und Tiere auf Wanderschaft. Katalog des Landesmuseums St. Pölten, pp 71–81
- Moog O, Leitner P, Huber T (2013) Makrozoobenthos. In: Ofenböck G (ed) Aquatische Neobiota in Österreich. Stand 2013. Bundesministerium f. Land- und Forstwirtschaft, Umwelt und Wasserwirtschaft, pp 54–92



- Moss B (2008) The Water Framework Directive: Total environment or political compromise? Sci. Total Environ. 400, 32–41. doi:10.1016/j.scitotenv.2008.04.029
- Navodaru I, Staras M, Cernisencu I (2001) The challenge of sustainable use of the Danube Delta Fisheries, Romania. Fisheries Management and Ecology 8: 323-332
- Nowak MM (2012) Zeitliche und räumliche Ausbreitungsmuster der invasiven Dreissena rostriformis bugensis Andrusov, 1897 in Deutschland und ihr Einfluss auf assoziierte Arten. Masterthesis, Institut für Tierökologie und spezielle Zoologie, Liebig-Universität Gießen, pp 1–63
- Orendt C, Schmitt C, Liefferinge C, Wolfram G, Deckere E (2009) Include or exclude? A review on the role and suitability of aquatic invertebrate neozoa as indicators in biological assessment with special respect to fresh and brackish European waters. Biol. Invasions 12, 265–283. doi:10.1007/s10530-009-9448-x
- Pasetti F (1862) Notizen über die Donauregulierung im Österreichischen Kaiserstaate bis zu Ende des Jahres 1861 mit Bezug auf die im k. k. Staatsministerium herausgegebenen Übersichts-Karte der Donau. Vienna, p 39
- Pauly D, Palomares ML (2005) Fishing down Marine Food Web: It is far more pervasive than we thought. Bulletin of Marine Science 76(2): 197-211.
- Petrović A, Simić V Milošević D, Paunović M. & Sivec I (2014) Diversity and Distributional Patterns of Stoneflies (Insecta: Plecoptera) in the Aquatic Ecosystems of Serbia (Central Balkan Peninsula). Acta zool. bulg., 66 (4): 517-526.
- Petts G & Amoros C (1996) Fluvial hydrosystems. Chapman & Hall
- Petts G.E., Möller H. & Roux A.L. (1989). Historical change of large alluvial rivers: Western Europe, John Wiley & Sons
- Pöckl M (2006) Strategies of a successful new invader in European fresh waters: fecundity and reproductive potential of the Ponto-Caspian amphipod Dikerogammarus villosus in the Austrian Danube, compared with the indigenous Gammarus fossarum and G. roeseli. Freshw Biol 52:50–63
- Poff NL (1997) Landscape filters and species traits: towards mechanistic understanding and prediction in stream ecology. J. North Am. Benthol. Soc. 16, 391–409.
- Poff NL, Ward JV (1990) Physical habitat template of lotic systems: Recovery in the context of historical pattern of spatiotemporal heterogeneity. Environ. Manage. 14, 629–645. doi:10.1007/BF02394714
- Popijac A (2008) Crveni popis obal~ara (Plecoptera) Hrvatske. Dr`avni zavod za za{titu prirode, Zagreb. [Red list of stoneflies (Plecoptera) of Croatia. State Institute for Nature Protection, Zagreb.] Available from http://www.dzzp.hr/eng_redlist.htm (accessed March 15, 2009).
- Popijac A & Sivec I (2009): Stoneflies (Insecta, Plecoptera) from museum collections in Croatia. NAT. CROAT. VOL. 18 No 2 243–254.
- Rahel FJ, Olden JD (2008) Assessing the effects of climate change on aquatic invasive species. Conserv Biol 22(3):521–533
- Rauser J (1960) Prispevek k limnickd zoogeografii Dunaje. Geograf 12:262–283
- Repassy N (1906) Wahrnehmung der Interessen der Fischerei beim Wasserbau. Stenographisches Protokoll über die Verhandlungen des Internationalen Fischerei-Kongresses. Verlag der k. k. österreichischen Fischerei-Gesellschaft, Wien
- Rood SB, Samuelson GM, Braatne JH, Gourley CR, Hughes FMR, Mahoney JM (2005) Managing river flows to restore floodplain forests. Frontiers in Ecology and Environment 3: 193-201.
- Roth NE, Allan, JD, Erickson DL (1996) Landscape influences on stream biotic integrity assessed at multiple spatial scales. Landsc. Ecol. 11, 141–156. doi:10.1007/BF02447513
- Ruffoni, A & Le Doaré J (2009): Nouvelles citations de presence d'Isogenus nubecula Newman, 1833 en France [Plecoptera, Perlodidae] Ephemera, 2008 (2009), Vol. 10 (2) : 95-102
- Russev B (1962) Die Insektenfauna der Donau vor dem bulgarischen Ufer. Mitteilungen der Versuchsstation für Süsswasserfischzucht, Plovdiv 1:115–128



- Schäffer JC (1775) Der Afterholzbock, vormals in einem lateinischen Sendschreiben an den Herrn von Reaumur etc., itzo in deutscher Sprache beschrieben, und mit einer Nachricht von der Frühlingsfliege mit kurzen Oberflügeln begleitet. E. A. Weiß, Regensburg
- Schiemer F, Guti G, Keckeis H, Staras M (2003) Ecological Status and problems of the Danube River and its fish fauna: A review. In: Welcomme R, Petr T (eds) Proceedings of the second International Symposium on the Management of Large Rivers for Fisheries: Sustaining Livelihoods and Biodiversity in the New Millennium, 11-14 February 2003, Phnom Penh, Kingdom of Cambodia. Volume I. Bangkok: Regional office for Asia and the Pacific, RAP Publications 2004/16: 273-299.
- Schiemer F, Zalewski, M (1992) The importance of riparian ecotones for diversity and productivity of riverine fish communities. Netherlands J. Zool. 42, 323–335. doi:10.1163/156854291X00360
- Schludermann E, Liedermann M, Hoyer H, Tritthart M, Habersack H, Keckeis H (2014) Effects of vesselinduced waves on the YOY-fish assemblage at two different habitat types in the main stem of a large river (Danube, Austria). Hydrobiologia 729(1):3–15
- Schmall B, Friedrich T (2014) Das Schicksal der großen Störarten in der Oberen Donau. Denisia 33 and Katalog des oberösterreichischen Landesmuseums Neue Serie 163: 423-442
- Schmidt-Kloiber A, Neu, PJ and Graf, W (2016). Metadata to the Distribution Atlas of European Plecoptera (in prep).
- Schmidt-Kloiber A, Vogl R and Hartmann A (2013) ECOPROF 4.0; available at: http://www.ecoprof.at/
- Schöll F (2006) Rhein-Messprogramm Biologie 2006/2007. Teil II-D. Das Makrozoobenthos des Rheins 2006/2007. Internationale Kommission zum Schutz des Rheins, Bericht 172:1–28
- Sivec, I. (2002) Rdei seznam vrbnic (Plecoptera) [Red list of stoneflies (Plecoptera) of Slovenia]. Priloga22
 Pravilnik o uvrstitvi ogro`enih rastlinskih in `ivalskih vrst v rde~i seznam. Uradni list Republike Slovenije 82, 8953–8953. Available from http://www.uradni- -list.si (accessed March 15, 2009)
- Staras M (1999) Fishery in relation to the environment in the Danube Delta Biosphere Reserve. In: Nijland HJ (ed) Dealing with nature in deltas. Lelystad, RIZA: 157-168
- Statzner B (1981) Shannon-Weaver diversity of the macrobenthos in the Schierensee brook (North Germany) and problems of its use for the interpretation of the community structure. Ver Internat Verein Limnol 21:782–786
- Statzner B, Bonada N, Doledec S (2008) Biological attributes discriminating invasive from native European stream macroinvertebrates. Biol Invasions 10:517–530
- Steffen W, Broadgate W, Deutsch L, Gaffney O, Ludwig C (2015) The trajectory of the Anthropocene: The Great Acceleration. Anthr. Rev. 2, 81–98. doi:10.1177/2053019614564785
- Travis JMJ (2003) Climate change and habitat destruction: a deadly anthropogenic cocktail. Proc R Soc Lond B 270:467–473
- Tittizer, T., Schöll, F., Dommermuth, M., Bäther, J., and Zimmer, M. (1991). Zur Bestandsentwicklung des Zoobenthos des Rheins im Verlauf der letzten neun Jahrzehnte, Wasser und Abwasser. 35:125 166 pp.
- Tobias W (1996) Sommernächtliche Schneetreiben am Main. Zum Phänomen des Massenfluges von Eintagsfliegen. Natur und Museum 126(2):37–54
- Tockner K., Robinson C.T., Uehlinger U. (2009) Rivers of Europe.
- Tockner K, Stanford JA (2002) Riverine floodplains: present state and future trends. Environmental Conservation 29: 308-330.
- Thorp JH, Thoms MC, Delong MD (2006) The riverine ecosystem synthesis: biocomplexity in river networks across space and time. River Res. Appl. 22, 123–147. doi:10.1002/rra.901
- Townsend CR, Downes BJ, Peacock K and Arbuckle CJ (2004) Scale and the detection of land-use effects on morphology, vegetation and macroinvertebrate communities of grassland streams. Freshwater Biology 49: 448–462.
- Townsend, CR (1989) The patch dynamics concept of stream community ecology. J. North Am. Benthol.



Soc. 8, 36–50.

- Townsend CR, Scarsbrook MR, Dolèdec S (1997) The intermediate disturbance hypothesis, refugia, and biodiversity in streams. Limnol. Oceanogr. 42, 938–949. doi:10.4319/lo.1997.42.5.0938
- Geest GJ van, Geerling GW & A.D. Buijse eds. (2015) Large river regulation and rehabilitation in Europe six selected case studies. REFORM deliverable 3.5 (FP7 Grant Agreement 282656)
- Vannote RL, Minshall GW, Sedell JR, Cushing CE (1980). The River Continuum Concept. Can. J. Fish. Aquat. Sci. 37, 130–137
- Verdú J.R. Y Galante E., eds. (2006) Libro Rojo de los Invertebrados de España. Dirección General para la Biodiversidad, Ministerio de Medio Ambiente, Madrid
- Vogt J, Soille P, Colombo R, Paracchini ML (2007). Development of a pan-European River and Catchment Database. Data Process., Lecture Notes in Geoinformation and Cartography 121–144Wallace, J.B. and Merrit, R.W. (1980) Filter-feeding ecology of aquatic insects. Annual Review of Entomolgy 25: 103–132
- Ward JV (1989). The four-dimensional nature of lotic ecosystems. J. North Am. Benthol. Soc. 8, 2–8
- Ward JV, Tockner K, Arscott D, Claret C (2002). Riverine landscape diversity. Freshw. Biol. 47, 517–539. doi:10.1046/j.1365-2427.2002.00893.x
- Wantzen, K.M. & N. Richard (2014) Mise en place d'un réseau d'Observation de la Biodiversité de la Loire et de ses Affluents (OBLA). Rapport scientifique Macroinvertébrés aquatiques: 59-69
- Wanzenböck J & Spindler T (1995) Rediscovery of Umbra krameri WALBAUM, 1792, in Austria and subsequent investigations (Pisces: Umbridae). Annalen des Naturhistorischen Museums Wien 97B: 450-457
- Wanzenböck J (1995) Current knowledge on the European mudminnow, Umbra krameri WALBAUM, 1792 (Pisces: Umbridae). Annalen des Naturhistorischen Museums Wien 97B: 439-449
- Wiens JA (2002) Riverine landscapes: taking landscape ecology into the water. Freshw. Biol. 47, 501–515. doi:10.1046/j.1365-2427.2002.00887.x
- Wiesner C (2005) New records of non-indigenous gobies (Neogobius spp.) in the Austrian Danube. Journal of Applied Ichthyology, 21, pp.324-327
- Wolf, B & T Widdig (2013): Rote Liste der Steinfliegen (Plecoptera) Hessens 2. Fassung. Hessisches Ministerium für Umwelt, Klimaschutz, Landwirtschaft und Verbraucherschutz (HMUKLV)
- Zwick P (1984) Mathamea beraudi (Nava's) and its European congeners (Plecoptera: Perlidae). Ann Limnol 20(1-2):129-139
- Zwick P (1992) Stream habitat fragmentation a threat to biodiversity. Biodivers Conserv 1:80-97
- Zwick P (1999) Historische Dokumente zur Fauna der Elbe bei Dresden vor hundert Jahren. Lauterbornia 37:97–112



Appendix

Fig. 30: Location of records (upper panel) and according large river catchments (lower panel) of Agnetina elegantula before 1990 (left) and after 1990 (right)



MARS PROJECT



Fig. 31: Location of records (upper panel) and according large river catchments (lower panel) of Besdolus ventralis before 1990 (left) and after 1990 (right)



MARS PROJECT



Fig. 32: Location of records (upper panel) and according large river catchments (lower panel) of Brachyptera braueri before 1990 (left) and after 1990 (right)





Fig. 33: Location of records (upper panel) and according large river catchments (lower panel) of Brachyptera trifasciata before 1990 (left) and after 1990 (right)





Fig. 34: Location of records (upper panel) and according large river catchments (lower panel) of Isogenus nubecula before 1990 (left) and after 1990 (right)





Fig. 35: Location of records (upper panel) and according large river catchments (lower panel) of Isoperla obscura before 1990 (left) and after 1990 (right)





Fig. 36: Location of records (upper panel) and according large river catchments (lower panel) of Isoperla pawlowskii before 1990 (left) and after 1990 (right)





Fig. 37: Location of records (upper panel) and according large river catchments (lower panel) of Marthamea vitripennis before 1990 (left) and after 1990 (right)



MARS PROJECT



Fig. 38: Location of records (upper panel) and according large river catchments (lower panel) of Perlodes dispar before 1990 (left) and after 1990 (right)



MARS PROJECT



Fig. 39: Location of records (upper panel) and according large river catchments (lower panel) of Xanthoperla apicalis before 1990 (left) and after 1990 (right)